## Maternal positions and mobility during first stage labour (Review)

Lawrence A, Lewis L, Hofmeyr GJ, Dowswell T, Styles C



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2009, Issue 4

http://www.thecochranelibrary.com



## TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
	6
	9
	10
ACKNOWLEDGEMENTS	11
	11
	15
DATA AND ANALYSES	41
Analysis 1.1. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 1 Duration of first stage of labour (hours): all women.	44
Analysis 1.2. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 2 Mode of	•
birth: spontaneous vaginal; all women	45
Analysis 1.3. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 3 Mode of birth: operative/assisted: all women.	46
Analysis 1.4. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care. Outcome 4 Mode of	
birth: caesarean section: all women.	48
Analysis 1.7. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 7 Maternal	(0
pain	49
pain.	50
	50
Analysis 1.10. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 10 Analgesia amount	52
Analysis 1.11. Comparison 1. Unright and ambulant positions versus recumbent positions and bed care. Outcome 11	)2
Duration of second stage of labour (minutes)	52
Analysis 1.12 Comparison 1. Unright and ambulant positions versus recumbent positions and bed care. Outcome 12	)2
Augmentation of labour using ovytocin	53
Analysis 1.13 Comparison 1 Unright and ambulant positions versus recumbent positions and bed care. Outcome 13	))
Artificial runture of membranes	54
Analysis 1.15 Comparison 1 Unright and ambulant positions versus recumbent positions and bed care. Outcome 15	74
Fstimated blood loss > 500 ml	54
Analysis 1.16 Comparison 1 Unright and ambulant positions versus recumbent positions and bed care. Outcome 16	71
Perineal trauma	55
Analysis 1.17 Comparison 1 Unright and ambulant positions versus recumbent positions and bed care. Outcome 17 Fetal	))
distress (requiring immediate delivery).	55
Analysis 1.19. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 19 Apgar	
scores	56
Analysis 1.20. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 20	
Admission to NICU	57
Analysis 1.21. Comparison 1 Upright and ambulant positions versus recumbent positions and bed care, Outcome 21	
Perinatal mortality	57
Analysis 2.1. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all	
women), Outcome 1 Duration of first stage of labour: time of epidural insertion to complete cervical dilation	
(minutes)	58
Maternal positions and mobility during first stage labour (Review)	—i
Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.	

Analysis 2.2. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 2 Mode of birth: spontaneous vaginal.
Analysis 2.3. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 3 Mode of birth: operative spontaneous/assisted
Analysis 2.4. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 4 Mode of birth: caesarean section
Analysis 2.6. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 6 Maternal pain
Analysis 2.7. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 7 Analgesia type
Analysis 2.8. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 8 Analgesia amount.
Analysis 2.9. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 9 Duration of second stage of labour (minutes).
Analysis 2.10. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women). Outcome 10 Augmentation of labour using oxytocin.
Analysis 2.12. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women), Outcome 12 Hypotension requiring intervention.
Analysis 2.17. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all
women). Outcome 17 Apgar scores
Analysis 3.1 Comparison 3 Standing and walking versus non-ambulant positions. Outcome 1 Duration of the first stage of
labour
HISTORY
$\begin{array}{c} \text{DECLARATIONS OF INTEREST} \\ \text{COUDCES OF SUBDOPT} \end{array}$
SUURCES OF SUFFORT
DIFFEKENCES BEI WEEN PROTOCOL AND REVIEW
INDEX IERMS

[Intervention Review]

## Maternal positions and mobility during first stage labour

Annemarie Lawrence<sup>1</sup>, Lucy Lewis<sup>2</sup>, G Justus Hofmeyr<sup>3</sup>, Therese Dowswell<sup>4</sup>, Cathy Styles<sup>5</sup>

<sup>1</sup>Institute of Women's and Children's Health (15), The Townsville Hospital, Douglas, Australia. <sup>2</sup>The School of Women's and Infants' Health/The School of Paediatrics and Child Health, The University of Western Australia, Subiaco, Australia. <sup>3</sup>Department of Obstetrics and Gynaecology, East London Hospital Complex, University of the Witwatersrand, University of Fort Hare, Eastern Cape Department of Health, East London, South Africa. <sup>4</sup>Cochrane Pregnancy and Childbirth Group, School of Reproductive and Developmental Medicine, Division of Perinatal and Reproductive Medicine, The University of Liverpool, UK. <sup>5</sup>Institute of Women's and Children's Health, The Townsville Hospital, Douglas, Australia

Contact address: Annemarie Lawrence, Institute of Women's and Children's Health (15), The Townsville Hospital, 100 Angus Smith Drive, Douglas, Queensland, 4810, Australia. annemarie\_lawrence@health.qld.gov.au. annielaw@bigpond.net.au. (Editorial group: Cochrane Pregnancy and Childbirth Group.)

Cochrane Database of Systematic Reviews, Issue 4, 2009 (Status in this issue: Unchanged) Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. DOI: 10.1002/14651858.CD003934.pub2 This version first published online: 15 April 2009 in Issue 2, 2009. Last assessed as up-to-date: 30 December 2008. (Help document - Dates and Statuses explained)

This record should be cited as: Lawrence A, Lewis L, Hofmeyr GJ, Dowswell T, Styles C. Maternal positions and mobility during first stage labour. *Cochrane Database of Systematic Reviews* 2009, Issue 2. Art. No.: CD003934. DOI: 10.1002/14651858.CD003934.pub2.

#### ABSTRACT

#### Background

It is more common for women in the developed world, and those in low-income countries giving birth in health facilities, to labour in bed. There is no evidence that this is associated with any advantage for women or babies, although it may be more convenient for staff. Observational studies have suggested that if women lie on their backs during labour this may have adverse effects on uterine contractions and impede progress in labour.

#### Objectives

The purpose of the review is to assess the effects of encouraging women to assume different upright positions (including walking, sitting, standing and kneeling) versus recumbent positions (supine, semi-recumbent and lateral) for women in the first stage of labour on length of labour, type of delivery and other important outcomes for mothers and babies.

#### Search strategy

We searched the Cochrane Pregnancy and Childbirth Group's Trials Register (November 2008).

#### Selection criteria

Randomised and quasi-randomised trials comparing women randomised to upright versus recumbent positions in the first stage of labour.

#### Data collection and analysis

We used methods described in the *Cochrane Handbook for Systematic Reviews of Interventions* for carrying out data collection, assessing study quality and analysing results. A minimum of two review authors independently assessed each study.

#### Main results

Maternal positions and mobility during first stage labour (Review) Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

The review includes 21 studies with a total of 3706 women. Overall, the first stage of labour was approximately one hour shorter for women randomised to upright as opposed to recumbent positions (MD -0.99, 95% CI -1.60 to -0.39). Women randomised to upright positions were less likely to have epidural analgesia (RR 0.83 95% CI 0.72 to 0.96). There were no differences between groups for other outcomes including length of the second stage of labour, mode of delivery, or other outcomes related to the wellbeing of mothers and babies. For women who had epidural analgesia there were no differences between those randomised to upright versus recumbent positions for any of the outcomes examined in the review. Little information on maternal satisfaction was collected, and none of the studies compared different upright or recumbent positions.

#### Authors' conclusions

There is evidence that walking and upright positions in the first stage of labour reduce the length of labour and do not seem to be associated with increased intervention or negative effects on mothers' and babies' wellbeing. Women should be encouraged to take up whatever position they find most comfortable in the first stage of labour.

### PLAIN LANGUAGE SUMMARY

#### Mothers' position during the first stage of labour

Women in the developed world and in health facilities in low-income countries usually lie in bed during the first stage of labour. Elsewhere, women progress through this first stage while upright, either standing, sitting, kneeling or walking around, although they may choose to lie down as their labour progresses. The attitudes and expectations of healthcare staff, women and their partners have shifted with regard to pain, pain relief and appropriate behaviour during labour and childbirth. A woman semi-reclining or lying down on the side or back during the first stage of labour may be more convenient for staff and can make it easier to monitor progression and check the baby. Fetal monitoring, epidurals for pain relief, and use of intravenous infusions also limit movement. Lying on the back (supine) puts the weight of the pregnant uterus on abdominal blood vessels and contractions may be less strong than when upright. Effective contractions help cervical dilatation and the descent of the baby.

The results of the review suggest that the first stage of labour may be approximately an hour shorter for women who are upright or walk around during the first stage of labour. The women's body position did not affect the rate of interventions. The review authors identified 21 controlled studies from a number of countries that randomly assigned a total of 3706 women to upright or recumbent positions in the first stage of labour. Nine of the studies included only women who were giving birth to their first baby. The length of the second stage of labour and the numbers of women who achieved spontaneous vaginal deliveries or required assisted deliveries and augmentation were similar between groups, where reported. Use of opioid analgesia was no different, although women randomised to upright positions were less likely to have epidural analgesia. In those studies specifically examining position and mobility for women receiving epidural analgesia (five trials, 1176 women), an upright or recumbent position did not change the length of the first stage of labour (time from epidural insertion to complete cervical dilatation) or rates of spontaneous vaginal, assisted and caesarean delivery. Little information was given on maternal satisfaction or outcomes for babies.

## BACKGROUND

In cultures not influenced by Western society, women progress through the first stage of labour in upright positions and change position as they wish with no evidence of harmful effects to either the mother or the baby (Andrews 1990; Gupta 2004; Roberts 1989). It is more common for women in the developed world to labour in bed (Boyle 2000; Roberts 1989; Simkin 1989). However, when these women are encouraged, they will choose a number of different positions as the first stage progresses (Carlson 1986; Fenwick 1987; Roberts 1989; Rooks 1999). Some studies have suggested that as a woman reaches five to six centimetres dilatation, there is a preference to lie down (Roberts 1980; Roberts 1984; Williams 1980). This may explain why women in randomised trials frequently have difficulty maintaining the position to which they have been assigned (Goer 1999), and suggests that there may not be a perfect universal position for women in the first stage of labour.

Recumbent (lying down) positions in the first stage of labour can have several practical advantages for the care provider; potentially making it easier to palpate the mother's abdomen to monitor contractions, perform vaginal examinations, check the baby's position, and listen to the baby's heart. Some developments in technology such as fetal monitoring, epidurals for pain relief and the use of intravenous infusions have all made it difficult and potentially unsafe for women to move about during labour.

Numerous studies have shown that a supine position in labour may have adverse physiological effects on the condition of the woman and her baby and on the progression of labour. The weight of the pregnant uterus can compress the abdominal blood vessels, compromising the mother's circulatory function including uterine blood flow (Abitbol 1985; Huovinen 1979; Marx 1982; Ueland 1969), and this may negatively affect the blood flow to the placenta (Cyna 2006; Roberts 1989; Rooks 1999; Walsh 2000). The effects of a woman's position on the frequency and intensity of contractions have also been examined (Caldeyro-Barcia 1960; Lupe 1986; Mendez-Bauer 1980; Roberts 1983; Roberts 1984; Ueland 1969). The findings indicated that contractions increased in strength in the upright or lateral position compared to the supine position and were often negatively affected when a labouring woman lay down after being upright or mobile. This effect can often be reversed if the woman returns to an upright position. Effective contractions are vital to aid cervical dilatation and fetal descent (Roberts 1989; Rooks 1999; Walsh 2000) and therefore have an important role in helping to reduce dystocia (slow progress in labour).

Moving about can increase a woman's sense of control in labour by providing a self-regulated distraction from the challenge of labour (Albers 1997). Support from another person also appears to facilitate normal labour (Hodnett 2007). Increasing a woman's sense of control may have the effect of decreasing her need for analgesia (Albers 1997; Hodnett 2007; Lupe 1986; Rooks 1999) and it has also been suggested that upright positions in the first stage of labour may increase women's comfort (Simkin 2002).

Because different groups advocate various positions in the first stage of labour, it seems particularly important to assess the available evidence so that positions which are shown to be safe and effective can be encouraged.

A related Cochrane review focuses on maternal position for fetal malpresentation in labour (Hunter 2007).

## OBJECTIVES

The purpose of this review is to assess the effects of different upright and recumbent positions and mobilisation for women in the first stage of labour on length of labour, type of delivery and other important outcomes for mothers and babies.

The primary objective is:

• to compare the effects of upright (defined as walking and upright non-walking, e.g. sitting, standing, kneel-

ing, squatting and all fours) positions with recumbent positions (supine, semi-recumbent and lateral) assumed by women in the first stage of labour on maternal, fetal and neonatal outcomes.

The secondary objectives are:

- to compare the effects of semi-recumbent and supine positions with lateral positions assumed by women in the first stage of labour on maternal, fetal and neonatal outcomes;
- to compare the effects of walking with upright nonwalking positions (sitting, standing, kneeling, squatting, all fours) assumed by women in the first stage of labour on maternal, fetal and neonatal outcomes;
- to compare the effects of walking with recumbent positions (supine, semi-recumbent and lateral) assumed by women in the first stage of labour on maternal, fetal and neonatal outcomes;
- to compare allowing women to assume the position/s they choose with recumbent positions (supine, semirecumbent and lateral) assumed by women in the first stage of labour on maternal, fetal and neonatal outcomes.

## METHODS

## Criteria for considering studies for this review

#### **Types of studies**

Randomised or quasi-randomised trials. We planned to include cluster randomised trials which were otherwise eligible. Cross-over trials might be useful for short-term outcomes such as fetal heart rate patterns, but would not be appropriate for the main outcomes of this review and were not included.

#### **Types of participants**

Women in the first stage of labour.

#### **Types of interventions**

The type of intervention was the position or positions assumed by women in the first stage of labour. The positions assumed by a women in the first stage of labour can be broadly categorised as being either upright or recumbent.

The positions considered recumbent were:

• semi recumbent;

Maternal positions and mobility during first stage labour (Review)

- lateral;
- supine.

The positions considered upright included:

- sitting;
- standing;
- walking;
- kneeling;
- squatting;
- all fours (hands and knees).

#### Types of outcome measures

Primary maternal outcomes:

- length of first stage of labour;
- type of delivery (spontaneous vaginal delivery, operative vaginal or caesarean);
- maternal satisfaction with positioning and with the childbirth experience.

Primary fetal and neonatal outcomes:

- fetal distress requiring immediate delivery;
- use of neonatal mechanical ventilation.

Secondary maternal outcomes:

- pain as experienced by the woman;
- use of analgesics (amount and type, e.g. epidural/opioid);
- length of second stage of labour;
- augmentation of labour using oxytocin;
- artificial rupture of membranes;
- spontaneous rupture of membranes;
- hypotension requiring intervention;
- estimated blood loss > 500 ml;
- perineal trauma (including episiotomy and third and fourth degree tears).

Secondary neonatal outcomes:

- · Apgar of less than seven at five minutes following deliverv;
- admission to the neonatal intensive care unit.

## Search methods for identification of studies

## **Electronic searches**

We searched the Cochrane Pregnancy and Childbirth Group's Trials Register by contacting the Trials Search Co-ordinator (31 December 2008).

The Cochrane Pregnancy and Childbirth Group's Trials Register is maintained by the Trials Search Co-ordinator and contains trials identified from:

- 1. quarterly searches of the Cochrane Central Register of Controlled Trials (CENTRAL);
- 2. weekly searches of MEDLINE;
- 3. handsearches of 30 journals and the proceedings of maior conferences;
- 4. weekly current awareness alerts for a further 44 journals plus monthly BioMed Central email alerts.

Details of the search strategies for CENTRAL and MEDLINE, the list of handsearched journals and conference proceedings, and the list of journals reviewed via the current awareness service can be found in the 'Specialized Register' section within the editorial information about the Cochrane Pregnancy and Childbirth Group.

Trials identified through the searching activities described above are each assigned to a review topic (or topics). The Trials Search Co-ordinator searches the register for each review using the topic list rather than keywords.

#### Searching other resources

We performed a manual search of the references of all retrieved articles and contacted expert informants.

We did not apply any language restrictions.

## Data collection and analysis

We used methods described in the Cochrane Handbook for Systematic Reviews of Interventions for data collection, assessing study quality and analysing results (Higgins 2008).

#### **S**election of studies

A minimum of two review authors independently assessed for inclusion all the potential studies identified as a result of the search strategy. We resolved any disagreement through discussion, or when required we consulted an additional person.

#### Data extraction and management

We designed a form to extract data. At least two review authors extracted the data using the agreed form. We resolved discrepancies through discussion, or if required we consulted a third author. We entered data into Review Manager software (RevMan 2008), and checked for accuracy.

When information regarding any of the above was unclear, we attempted to contact authors of the original reports to provide further details.

#### Assessment of risk of bias in included studies

Two review authors independently assessed risk of bias for each study using the criteria outlined in the Cochrane Handbook for Systematic Reviews of Interventions (Higgins 2008). We resolved any disagreement by discussion or by involving a third assessor.

Maternal positions and mobility during first stage labour (Review)

Please see the 'Risk of bias' tables following the Characteristics of included studies tables for the assessment of bias for each study.

## (1) Sequence generation (checking for possible selection bias)

We described for each included study the methods used to generate the allocation sequence to assess whether methods were truly random.

We assessed the methods as:

- adequate (e.g. random number table; computer random number generator);
- inadequate (odd or even date of birth; hospital or clinic record number); or
- unclear.

## (2) Allocation concealment (checking for possible selection bias)

We described for each included study the method used to conceal the allocation sequence in sufficient detail and determined whether group allocation could have been foreseen in advance of, or during, recruitment, or changed afterwards.

We have assessed the methods as:

- adequate (e.g. telephone or central randomisation; consecutively numbered sealed opaque envelopes);
- inadequate (open random allocation; unsealed or nonopaque envelopes, alternation; date of birth);
- unclear.

#### (3) Blinding (checking for possible performance bias)

We have described for each included study the methods used to blind study personnel from knowledge of which intervention a participant received. We have described where there was any attempt at partial blinding (e.g. of outcome assessors). It is important to note that with the types of interventions described in this review, blinding participants to group assignment is generally not feasible. Similarly, blinding staff providing care is very difficult, and this may have the effect of increasing co-interventions, which in turn may affect outcomes. The lack of blinding in these studies may be a source of bias, and this should be kept in mind in the interpretation of results.

We assessed the methods as:

- adequate, inadequate or unclear for participants;
- adequate, inadequate or unclear for personnel;
- adequate, inadequate or unclear for outcome assessors.

(4) Incomplete outcome data (checking for possible attrition bias through withdrawals, dropouts, protocol deviations)

We have described for each included study the completeness of outcome data, including attrition and exclusions from the analysis. We state whether attrition and exclusions were reported, the numbers (compared with the total randomised participants), reasons for attrition/exclusion where reported, and any re-inclusions in analyses which we have undertaken.

We assessed the methods as:

- adequate (e.g. where there was no missing data or low levels (less than 10%) and where reasons for missing data were balanced across groups);
- inadequate (e.g. where there were high levels of missing data (more than 10%));
- unclear (e.g. where there was insufficient reporting of attrition or exclusions to permit a judgement to be made).

#### (5) Other sources of bias and overall risk of bias

We described for each included study any important concerns we had about other possible sources of bias.

We have made explicit judgements about risk of bias for important outcomes both within and across studies. With reference to 1-4 above, we assessed the likely magnitude and direction of the bias and whether we considered it was likely to impact on the findings. We have explored the impact of risk of bias through undertaking sensitivity analyses; see Sensitivity analysis below.

### Measures of treatment effect

We carried out statistical analysis using the Review Manager software (RevMan 2008). We used fixed-effect meta-analysis for combining data in the absence of significant heterogeneity if trials were sufficiently similar. When significant heterogeneity was present, we used a random-effects meta-analysis.

#### Dichotomous data

For dichotomous data, we have presented results as summary risk ratio (RR) with 95% confidence intervals (CI).

#### Continuous data

For continuous data (e.g. maternal pain and satisfaction when measured as scores or on visual analogue scales) we have used the mean difference (MD) if outcomes were measured in the same way between trials. We used the standardised mean difference (SMD) to combine trials that measured the same outcome, but used different methods.

#### Unit of analysis issues

#### **Cluster-randomised trials**

We intended to include cluster-randomised trials in the analyses along with individually randomised trials, and to adjust sample sizes using the methods described in Gates 2005 and Higgins 2008.

Maternal positions and mobility during first stage labour (Review)

Copyright  $\textcircled{\sc c}$  2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

We identified no cluster randomised trials in this version of the review, but if we identify such trials in future searches we will include them in updates.

#### Dealing with missing data

For included studies, we noted levels of attrition. Where data were not reported for some outcomes or groups we attempted to contact the study authors for further information.

#### Intention to treat analysis (ITT)

We had intended to analyse data on all participants with available data in the group to which they were allocated, regardless of whether or not they received the allocated intervention. If in the original reports participants were not analysed in the group to which they were randomised, and there was sufficient information in the trial report, we have attempted to restore them to the correct group (e.g. we did this for the data from the Calvert 1982 study).

#### Assessment of heterogeneity

We examined heterogeneity using the I<sup>2</sup> statistic. Where we have identified high levels of heterogeneity among the trials (greater than 50%), we explored it by pre-specified subgroup analysis and by performing sensitivity analysis. A random-effects meta-analysis was used as an overall summary for these comparisons.

#### Subgroup analysis and investigation of heterogeneity

Where data were available, we had planned subgroup analyses by: - nulliparous versus multiparous women.

However, several trials recruited only nulliparous women, and in other trials results were presented separately for nulli- and multiparous women and no overall findings (for all women irrespective of parity) were reported. For example, for the primary review outcome (duration of the first stage of labour) of the nine trials providing data, four provided mean figures for nulli- and multiparous women, but no overall mean. Thus, for pragmatic reasons (in order to use all available data from trials) we have reported overall results for all women, but in the analysis data have been grouped according to parity if this is how data were presented in the trial reports.

We had also planned subgroup analysis by:

- women with a low-risk pregnancy (no complications, greater than or equal to 37 weeks' gestation, singleton with a cephalic presentation) versus high-risk pregnancy.

Data were not available to carry out this analysis.

#### Sensitivity analysis

We carried out sensitivity analyses to explore the effect of trial quality for important outcomes in the review. Where there was risk of bias associated with a particular aspect of study quality (e.g. inadequate allocation concealment or high levels of attrition), we explored this by sensitivity analysis.

## RESULTS

#### **Description of studies**

See: Characteristics of included studies; Characteristics of excluded studies.

#### **Results of the search**

We identified a total of 53 reports representing 47 studies by the search strategy.

#### **Included studies**

We included 21 studies with a total of 3706 women in the review. Studies were carried out in a number of countries; seven in the UK (Broadhurst 1979; Calvert 1982; Collis 1999; Fernando 1994; Flynn 1978; McManus 1978; Williams 1980); five in the USA (Andrews 1990; Bloom 1998; Mitre 1974; Nageotte 1997; Vallejo 2001); two in France (Frenea 2004; Karraz 2003); and one each in Finland (Haukkama 1982;), Sweden (Bundsen 1982), Hong Kong (Chan 1963), Japan (Chen 1987), Australia (MacLennan 1994), Brazil (Miquelutti 2007) and Thailand (Phumdoung 2007). Several of the studies included only nulliparous women (Andrews 1990; Chan 1963; Collis 1999; Fernando 1994; Miquelutti 2007; Mitre 1974; Nageotte 1997; Phumdoung 2007; Vallejo 2001). We have set out details of inclusion and exclusion criteria for individual studies and descriptions of the interventions in the Characteristics of included studies tables.

#### **Excluded studies**

We excluded 25 studies from the review. Several of the studies were not randomised trials or it was not clear that there had been random allocation to groups (Allahbadia 1992; Asselineau 1996; Caldeyro-Barcia 1960; Solano 1982); two of the studies used cross-over designs (Melzack 1991; Molina 1997). One study was excluded because the rate of attrition meant that it was difficult to interpret results: in the Diaz 1980 study, more than 30% of the intervention group were excluded post-randomisation because they did not comply with the protocol. In the Hemminki 1983 study, women in the two study groups received different packages of care, so it was not possible to separate out the possible treatment effect of maternal position on outcomes. The McCormick 2007 study had not taken place.

Maternal positions and mobility during first stage labour (Review)

In some studies, the intervention was not comparing mobility or upright positions with recumbent positions; for example, Cobo 1968 and Wu 2001 examined lying in bed on one side rather than the other, or lying supine. In some studies position/mobility was compared with a different type of intervention, for example the Hemminki 1985 study included women experiencing delay in labour and compared immediate oxytocin with ambulation and delayed oxytocin. Similarly, Read 1981 examined oxytocin in labour. The COMET 2001 study compared women receiving different types of epidural, while in the Hodnett 1982 study the main focus was on electronic fetal monitoring, and ambulation was an outcome rather than part of the intervention. Three studies focused on interventions in the second, rather than in the first stage of labour (Hillan 1984; Liu 1989; Radkey 1991).

Several studies, which may otherwise have been eligible, focused on outcomes which had not been pre-specified in this review. For example, Danilenko-Dixon 1996 focused on cardiac output, while the study by Schmidt 2001 and those by Ahmed 1985, Cohen 2002 and Schneider-Affeld 1982 (reported in brief abstracts) did not provide sufficient information on outcomes or present outcome data in a form that we were able to use in the review.

#### **Risk of bias in included studies**

The overall quality of the included studies was difficult to assess as many of the studies gave very little information about the methods used.

#### Allocation

The method of sequence generation was often not mentioned in the included studies. In the studies by Miquelutti 2007 and Vallejo 2001, a computer generated list of random numbers was used; five of the included studies utilised a quasi-randomised design, where the allocation to groups was according to hospital or casenote number or by alternate allocation (Calvert 1982; Chan 1963; Chen 1987; Haukkama 1982; Williams 1980); for the rest, the method of sequence generation was not stated.

The methods used to conceal group allocation from those recruiting women to the trials were also frequently not described. Six studies referred to group allocation details being contained in envelopes; in the studies by Collis 1999, MacLennan 1994 and Miquelutti 2007 the envelopes were described as sealed and opaque, and in the other studies envelopes were variously described as plain, numbered or sealed (Frenea 2004; McManus 1978). In sensitivity analysis where studies of better and poorer quality have been separated, we regarded the six studies which give details of allocation concealment as the better quality studies, while we regarded those studies where allocation concealment was inadequate (e.g. in the quasi-randomised studies) or where methods were unclear as of poorer quality.

#### Blinding

In interventions of this type, blinding women and clinical staff to group allocation is not generally feasible. It is possible to have partial blinding of outcome assessors for some types of outcomes, but it was not clear that this was achieved in any of the included studies. The lack of blinding may introduce bias, and this should be kept in mind in the interpretation of the results.

#### Follow up and exclusions

Details regarding loss to follow up are set out in the risk of bias tables. In general, loss to follow up was not a serious problem in these studies, as many of the outcomes were recorded during labour.

In one study (Chen 1987), there was a high level of post-randomisation exclusion in both study groups (37%). This study was also at high risk of bias because of poor allocation concealment. A sensitivity analysis was carried out to examine the effects on results of excluding this study, along with those others at high risk of bias for poor or unclear allocation concealment.

In one study we did not use the whole sample in the analyses. In the study by Phumdoung 2007, women were randomised into five separate groups (see Characteristics of included studies for a description of the groups); we selected the two groups which we thought best represented upright and recumbent positions to include in the analyses.

#### Other potential sources of bias

There was wide variation in the types of interventions tested in these studies. Some authors gave very little information on the intervention, for example at what stage in labour it was started, what exactly women were asked to do and what instructions were given to women in the control groups. This lack of detail means that the interpretation of results is not simple. Further, co-interventions in included studies also varied. Readers should bear this variability in mind when reading the results of the review.

#### **Effects of interventions**

Upright positions (including sitting, standing, walking and kneeling) versus recumbent positions - 16 trials, 2530 women

#### **Duration of labour**

#### Duration of the first stage of labour

The duration of the first stage of labour varied considerably within and between trials. There were high levels of heterogeneity when

Maternal positions and mobility during first stage labour (Review)

studies were pooled ( $I^2 = 79\%$ ). Hence, results need to be interpreted with caution, and in view of high levels of heterogeneity, we have used a random effects model for these analyses.

Overall, for all women, the first stage of labour was approximately one hour shorter for those randomised to upright compared with supine and recumbent positions; this analysis included pooled results from nine trials (including 1677 women) and the difference between groups was statistically significant (MD -0.99, 95% CI -1.60 to -0.39) (Analysis 1.1).

For nulliparous women, the length of the first stage of labour was not significantly different between groups; for multiparous women, the duration of first stage was approximately half an hour shorter for those randomised to upright positions, but the evidence of a difference between groups did not reach statistical significance.

#### Duration of the second stage of labour

There was no difference between groups in the length of the second stage of labour in the two trials that reported this outcome ( Analysis 1.11).

#### Mode of birth

#### Spontaneous vaginal birth

Results were similar for women randomised to upright versus recumbent positions, and this finding applied to both nulli- and multiparous women. There were no significant differences between groups in the numbers of women achieving spontaneous vaginal deliveries (Risk ratio (RR) 1.01, 95% confidence interval (CI) 0.97 to 1.05) (Analysis 1.2).

#### Operative spontaneous or assisted delivery

Women randomised to upright positions had similar rates of assisted deliveries compared with those randomised to recumbent positions (Analysis 1.3). Again, these results applied irrespective of parity.

#### Caesarean delivery

Women encouraged to maintain upright positions had slightly lower rates of caesarean section compared with those in comparison groups. However, the strength of evidence was weak, and results did not reach statistical significance (overall RR 0.73, 95% CI 0.51 to 1.07) (Analysis 1.4).

#### Maternal satisfaction

While some studies collected information on satisfaction with specific aspects of care (e.g. satisfaction with pain relief), we were not able to pool results, as none of these studies collected information on women's satisfaction with their general experience of childbirth.

#### Maternal pain and analgesia

There were no differences identified between groups in terms of reported discomfort or requests for analgesia, although relatively few trials examined these outcomes, and findings were inconsistent. Most studies collected information on the types of analgesia women received. There were no differences between groups in terms of use of opioid analgesia, although women randomised to upright positions were less likely to have epidural analgesia, and this difference reached statistical significance (RR 0.83 95% CI 0.72 to 0.96, P = 0.01). The amount of analgesia received by women in the two groups was measured in one trial, but the difference between groups was not statistically significant (Analysis 1.10).

#### Interventions in labour

#### Augmentation of labour using oxytocin

Women randomised to upright versus recumbent positions had similar rates of augmentation of labour (Analysis 1.12). In two studies, amniotomy was carried out routinely on all women included (Chen 1987; MacLennan 1994); one study examined differences in amniotomy rates in women allocated to upright compared with recumbent positions. There were no differences between groups (Haukkama 1982).

#### Maternal outcomes

Few studies reported maternal outcomes, so there was very little information on rates of post-partum haemorrhage and perineal trauma. Results from single trials suggest no significant differences between groups.

#### Fetal and neonatal outcomes

Again, there was little information from included studies on outcomes for babies. There were no significant differences between groups in terms of fetal distress and neonatal Apgar scores. Admission to special care units was only reported in one study and was slightly more likely for babies born to mothers randomised to upright positions, but this difference did not reach statistical significance (Analysis 1.20. One study examined perinatal deaths; no deaths in either group were recorded (Bloom 1998)

## Upright (including walking) versus recumbent positions - with epidural (five trials, 1176 women)

Analysis for this comparison is for all women, irrespective of parity. We had planned subgroup analysis by parity; however, of the five trials contributing data, three recruited nulliparous women only (Collis 1999; Nageotte 1997; Vallejo 2001) and the remaining two studies did not report results separately for nulliparous and multiparous women (Frenea 2004; Karraz 2003).

#### Duration of labour

#### Duration of the first stage of labour

There were no differences between groups in terms of the length of the first stage of labour (i.e. time from epidural insertion to complete cervical dilatation) (Analysis 2.1).

#### Mode of delivery

Rates of spontaneous vaginal, assisted and caesarean delivery were similar for women randomised to upright versus recumbent positions (Analysis 2.2; Analysis 2.3; Analysis 2.4).

#### Maternal pain, satisfaction and other outcomes

There were no statistically significant differences between groups in terms of maternal satisfaction, women receiving oxytocin augmentation, women experiencing hypotension, women requiring additional analgesia, or the amount of analgesia women received (Analysis 2.5 to Analysis 2.14). However, few trials measured these outcomes and results are based on results from only one or two studies.

#### **Neonatal outcomes**

There was no information on perinatal mortality or admission to special care units (Analysis 1.20; Analysis 2.18). There were no differences between groups in the incidence of Apgar scores of less than seven at one and five minutes (Analysis 2.17).

## Trials where ambulation was encouraged and trials where women were confined to bed or sitting

In order to address the question of whether standing and walking, rather than sitting or upright bed positions were associated with shorter length of labour, we carried out further analysis. In this analysis (Analysis 3.1) the majority of trials where women were encouraged to get out of bed and ambulate were analysed separately from the three trials (Chen 1987; Mitre 1974; Phumdoung 2007) where women were encouraged to sit or maintain nonambulant positions. In this analysis we have only included nulliparous women as two of the three trials examining non-ambulant positions (Mitre 1974; Phumdoung 2007) only recruited such women, and the third provided separate data for these women. Results suggest that non-ambulant upright positions (sitting in bed or on a sofa, or semi-kneeling in bed) were associated with shorter labours compared with comparison groups (MD -1.92 95% CI -2.83 to -1.01) whereas, for studies examining ambulation, the difference between intervention and comparison groups was not significant (MD -0.20 95% CI -1.36 to 0.96). For each of these comparisons, the level of heterogeneity was high and we have used a random effects meta-analysis. Overall, I<sup>2</sup> was 83% and there were differences in the direction of findings, and in the size of the treatment effect, so these results should be viewed with caution. As well as statistical heterogeneity, we suspected clinical heterogeneity, as there was wide variation in the mean length of labour in different studies. Further, the three studies where women maintained nonambulant positions examined different types of intervention in different settings. All the women in the study by Mitre 1974 had amniotomies and were confined to bed. Women in the study by Chen 1987 were provided with a settee and encouraged to sit on it, but could walk around if they wished to. In the Phumdoung 2007 study, women alternated between a semi-kneeling position and a semi-recumbent position. All of this variation means that it is difficult to draw firm conclusions about the most favourable positions for women to adopt.

#### Subgroup analysis

#### Low- and high-risk groups

Data were not available to carry out this analysis.

#### Sensitivity analysis

For the primary review outcomes, we carried out a sensitivity analysis whereby those trials with poor allocation concealment (e.g. alternate group allocation) or where no information on allocation concealment had been provided, were taken out of the meta-analysis to see if this would change the direction of results or the size of the effect. For duration of length of first stage, only one trial was left when trials with a high risk of bias were removed. In this trial there were no significant differences in duration of the first stage of labour between the ambulant and comparison groups, irrespective of parity (MD -0.25, 95% CI -1.68 to 1.18) (McManus 1978). When all trials were included results had suggested a shorter duration of first stage for those women in the intervention groups. For mode of delivery, three trials were included (MacLennan 1994; McManus 1978; Miquelutti 2007). Here, there were no significant differences between groups in terms of spontaneous vaginal, assisted vaginal or caesarean births. This finding was similar to that resulting from the inclusion of all trials.

## DISCUSSION

The objectives of this review were to assess the effects of positions and mobility during the first stage of labour on length of labour, type of delivery and other important outcomes for mothers and babies.

Women who were upright or mobile had a shorter first stage of labour compared with women who were supine (MD -0.99, 95% CI -1.60 to -0.39). Shorter length of labour is an important outcome, as every contraction is potentially painful. Women randomised to upright positions were also less likely to have epidural analgesia. However, there was little evidence that position or mobility had any effect on the rate of other interventions or on the wellbeing of mothers and babies.

When considering the results from the review, it is worth noting that designing trials to examine interventions in this area is challenging, and it is difficult to avoid bias. It is not possible to blind women and caregivers to group allocation. In addition, it is difficult to standardise interventions. For the trials included in the review, there was considerable variation in the interventions women received. Even where interventions appeared similar in different studies, it is likely that women's experience varied; this sort of intervention cannot be easily controlled. Women may have had difficulty maintaining the intervention position or preferred alternative positions. There was also variation in caregiver behaviour in relation to study protocols; in some studies women were strongly encouraged by staff to mobilise (e.g. in the study by Miquelutti 2007, any woman in the intervention group that remained in bed for more than 30 minutes was asked to get out again); in other studies, women had more choice and more gentle encouragement. In one study the intervention was only encouraged during the day as it was not felt that women would like to walk around at night (Karraz 2003), and in this same study, women in the comparison group were not allowed out of bed even to walk to the toilet. Further, there was huge variability in the amount of time women actually followed the protocol in terms of ambulation or staying in bed. For example, in the Calvert 1982 study, less than half of the women in the intervention group chose to get out of bed at all, and those that did get out, only tended to do so for short periods of time. .

Heterogeneity in study findings also created problems in interpreting results. For the main outcome - length of the first stage of labour - there was considerable variation within and between studies in terms of group means. Various studies defined and measured the length of the first stage of labour in different ways. Measurement may have commenced on admission or at various points of cervical dilatation according to different hospital policies or study designs.

We were not able to answer several of the questions set out in the protocol. There were no studies comparing different types of upright position, e.g. sitting up in bed or on a chair versus walking or kneeling, or other upright positions. Results suggest that nonambulant upright positions may reduce the length of labour, but only three studies (all with a high risk of bias) examined nonwalking positions and results were difficult to interpret because of the variability of interventions.

Few of the studies collected outcome data on many of the review outcomes such as pain, maternal satisfaction, and neonatal outcomes. Most of the included studies collected information on mode of birth, but few had the statistical power to detect differences between groups.

Studies were carried out over a long period: from the early 1960s (Chan 1963) through to 2007 (Miquelutti 2007; Phumdoung 2007); and in a number of different healthcare settings. The cul-

tural and healthcare context is likely to have been different at different times and in different settings, and there have also been changes in healthcare technologies. Within these changing contexts, the attitudes and expectations of healthcare staff, women and their partners towards pain, pain relief and appropriate behaviour during labour and childbirth have shifted. All of these factors are important in the interpretation of results.

This review needs to be looked at alongside other related Cochrane reviews focusing on care during labour (e.g. Cluett 2002; Gupta 2004; Hodnett 2007; Hunter 2007). While position in the first stage of labour may have an independent effect, the position in second stage and other variables (e.g. the presence of a birth companion) are also important.

## AUTHORS' CONCLUSIONS

#### Implications for practice

Upright positions and walking are associated with a reduction in the length of the first stage of labour, and women randomised to upright positions may be less likely to have epidural analgesia, but there was little evidence of differences for other maternal and infant outcomes. Despite the limited evidence from trials included in the review, observational studies suggest that maintaining a supine position in labour may have adverse physiological effects on the woman and her baby (Abitbol 1985; Huovinen 1979; Marx 1982; Roberts 1989; Rooks 1999; Walsh 2000). Therefore, women should be encouraged to take up whatever position they find most comfortable while avoiding spending long periods supine. Women's preferences may change during labour. Many women may choose an upright or ambulant position in early first stage labour and choose to lie down as their labour progresses.

## Implications for research

Overall, the quality of the studies included in the review was mixed and most studies provided little information on methods. Minimising risk of bias in trials on this topic is challenging, as blinding is not feasible and it is difficult to standardise interventions. At the same time, some aspects of study design can be controlled.

Some considerations for future research are as follows.

- There is a need for high-quality trials in this area, with particular attention given to allocation concealment.
- Trialists should clearly explain how they have defined the first stage of labour.
- There is a need to improve and standardise measurement of pain.
- There is an urgent need to collect information on outcomes for mothers, such as satisfaction with the experi-

ence of childbirth, and more information is needed on pain and the effect of position on complications such as haemorrhage.

- Few trials assessed outcomes for babies and future studies need to focus on this.
- Studies are needed which compare different upright positions (e.g. sitting upright versus walking) and different lying positions (e.g. lying on side versus back).

## ACKNOWLEDGEMENTS

We would like to thank Philippa Middleton, Caroline Crowther, Lea Budden and Joan Webster for their advice on early versions of this review.

As part of the pre-publication editorial process, this review has been commented on by two peers (an editor and referee who is external to the editorial team), a member of the Pregnancy and Childbirth Group's international panel of consumers and the Group's Statistical Adviser.

#### REFERENCES

#### References to studies included in this review

#### Andrews 1990 {published data only}

Andrews CM, Chrzanowski M. Maternal position, labour and comfort. *Applied Nursing Research* 1990;**3**(1):7–13.

#### Bloom 1998 {published data only}

Bloom SL, Kelly MA, Beimer H, Garcia M, Burpo B, McIntire DD, et al.A randomized trial of the effects of ambulation on active labor. *American Journal of Obstetrics and Gynecology* 1998;**178**(1 Pt 2):S98. \* Bloom SL, McIntire DD, Kelly MA, Beimer HL, Burpo RH, Garcia MA, et al.Lack of effect of walking on labor and delivery. *New England Journal of Medicine* 1998;**339**(2):76–9.

Goer H. Does walking enhance labor progress?. *Birth* 1999;**26**(2): 127–9.

#### Broadhurst 1979 {published data only}

Broadhurst A, Flynn AM, Kelly J, Lynch PF. The effect of ambulation in labour on maternal satisfaction, analgesia and lactation. Proceedings of 5th International Congress on Psychosomatic Medicine in Obstetrics and Gynaecology, "Emotion and Reproduction";1977 Nov 13-19; Rome, Italy. 1979:943–6.

#### Bundsen 1982 {published data only}

Bundsen P, Lundberg J, Peterson LE. Telemetric versus conventional fetal monitoring in labour - a prospective randomized study [abstract]. Proceedings of 8th European Congress of Perinatal Medicine; 1982 Sept 7-10; Brussels, Belgium. 1982:Abstract no: 256.

#### Calvert 1982 {published data only}

Calvert JP, Newcombe RG, Hibbard BM. An assessment of radiotelemetry in the monitoring of labour. *British Journal of Obstetrics and Gynaecology* 1982;**89**:285–91.

#### Chan 1963 {published data only}

Chan DPC. Positions during labour. *British Medical Journal* 1963; 1:100–2.

#### Chen 1987 {published data only}

Chen SZ, Aisaka K, Mori H, Kigawa T. Effects of sitting position on uterine activity during labor. *Obstetrics & Gynecology* 1987;**69**(1): 67–73.

#### Collis 1999 {published data only}

Collis R, Harding S, Davies L, Moore C, Baxendall M, Morgan B. Ambulation in labour with an epidural: the effect on analgesic requirements, outcome of labour and maternal satisfaction [abstract]. *International Journal of Obstetric Anesthesia* 1994;**3**:109.

\* Collis RE, Harding SA, Morgan BM. Effect of maternal ambulation on labour with low-dose combined spinal-epidural analgesia. *Anaesthesia* 1999;**54**(6):535–9.

#### Fernando 1994 {published data only}

Fernando R, Gill P, Urquhart J, Morgan BM. Neurobehavioural changes in neonates after ambulatory combined spinal epidural (CSE) analgesia during labour. *International Journal of Obstetric Anesthesia* 1994;**3**:174.

#### Flynn 1978 {published data only}

Flynn AM, Hollins G, Kelly J, Lynch PF. The effect of ambulation in labour on uterine action, analgesia and fetal well-being. Proceedings

of 6th European Congress of Perinatal Medicine; 1978 Aug 29-Sept 1; Vienna, Austria. 1978:Abstract no: 344.

\* Flynn AM, Kelly J, Hollins G, Lynch PF. Ambulation in labour. BMJ 1978;**2**:591–3.

#### Frenea 2004 {published data only}

Frenea S, Chirossel C, Rodriguez R, Baguet JP, Racinet C, Payen JF. The effects of prolonged ambulation on labor with epidural analgesia. *Anesthesia & Analgesia* 2004;**98**(1):224–9.

#### Haukkama 1982 {published data only}

Haukkama M, Purhonen M, Teramo K. The monitoring of labor by telemetry. *Journal of Perinatal Medicine* 1982;**10**:17–21.

#### Karraz 2003 {published data only}

Karraz MA. Ambulatory epidural anaesthesia and the duration of labor. *International Journal of Gynecology & Obstetrics* 2003;**80**:117–22.

#### MacLennan 1994 {published data only}

MacLennan AH, Crowther C, Derham R. Does the option to ambulate during spontaneous labour confer any advantage or disadvantage?. *Journal of Maternal-Fetal Medicine* 1994;**3**:43–8.

#### McManus 1978 {published data only}

McManus TJ, Calder AA. Upright posture and the efficiency of labour. *Lancet* 1978;1(8055):72-4.

#### Miquelutti 2007 {published data only}

Miquelutti MA, Cecatti JG, Makuch MY. Upright position during the first stage of labor: a randomised controlled trial. *Acta Obstetricia et Gynecologica Scandinavica* 2007;**86**(5):553–8.

#### Mitre 1974 {published data only}

Mitre IN. The influence of maternal position on duration of the active phase of labor. *International Journal of Gynecology & Obstetrics* 1974;**12**(5):181–3.

#### Nageotte 1997 {published data only}

Nageotte M, Larson D, Rumney P, Sidhu M, Hollenback K. A prospective randomized study of intrapartum epidural vs combination intrathecal/epidural anesthesia with or without ambulation. *American Journal of Obstetrics and Gynecology* 1997;**176**(1 Pt 2):S22. \* Nageotte MP, Larson D, Rumney PJ, Sidhu M, Hollenbach K. Epidural analgesia compared with combined spinal-epidural analgesia during labor in nulliparous women. *New England Journal of Medicine* 1997;**337**(24):1715–9.

### Phumdoung 2007 {published data only}

Phumdoung S, Youngvanichsate S, Jongpaiboonpatana W, Leetanaporn R. The effects of the PSU Cat position and music on length of time in the active phase of labor and labor pain. *Thai Journal of Nursing Research* 2007;**11**(2):96–105.

#### Vallejo 2001 {published data only}

Vallejo M, Firestone L, Mandell G, Jaime F, Makishima S, Ramanathan S. The effect of sitting and ambulating on labor duration and maternal outcome [abstract]. *Anesthesiology* 2001;**94**(1A): Abstract no: A7.

Vallejo M, Mandell G, Jaime F, Ramanathan S. Ropivacaine for walking epidural analgesia during labor. *Regional Anesthesia and Pain Medicine* 1999;**24**(3 Suppl):74.

\* Vallejo MC, Firestone LL, Mandell GL, Jaime F, Makishima S, Ramanathan S. Effect of epidural analgesia with ambulation on labor duration. Anesthesiology 2001;95(4):857-61.

Vallejo MC, Mandell GL, Jaime F, Makishima S, Ramanathan S. Walking epidural analgesia: the effect of ambulation on labor duration and maternal outcome [abstract]. *Anesthesiology* 2000;**93**(3A): Abstract no: A1069.

#### Williams 1980 {published data only}

Williams RM, Thom MH, Studd JWW. A study of the benefits and acceptability of ambulation in spontaneous labour. *British Journal of Obstetrics and Gynaecology* 1980;**87**:122–6.

#### References to studies excluded from this review

#### Ahmed 1985 {published data only}

Ahmed LT, Bouchetara K. The influence of maternal position on duration of labor [abstract]. *Archives of Gynecology* 1985;**237 Suppl**: 9.

#### Allahbadia 1992 {published data only}

Allahbadia GN, Vaidya PR. Why deliver in the supine position?. *Australian and New Zealand Journal of Obstetrics and Gynaecology* 1992;**32**(2):104–6.

#### Asselineau 1996 {published data only}

Asselineau D. Does ambulation under epidural analgesia during labour modify foetal extraction conditions? [La deambulation sous peridurale lors du travail modifie–t–elle les conditions d'extraction foetale?]. *Contraception, Fertilité, Sexualité* 1996;**24**(6):505–8.

#### Caldeyro-Barcia 1960 {published data only}

Caldeyro-Barcia R, Noriega-Guerra L, Cibils LA, Alvarez H, Poseiro JJ, Pose SV, et al.Effect of position changes on the intensity and frequency of uterine contractions during labor. *American Journal of Obstetrics and Gynecology* 1960;**80**(2):284–90.

#### Cobo 1968 {published data only}

Cobo E, De Bernal MM, Quintero CA, Cuadrado E. Neurohypophyseal hormone release in the human. III. Experimental study during labor. *American Journal of Obstetrics and Gynecology* 1968;**101**:479– 89.

#### Cohen 2002 {published data only}

Cohen S, Ayers C, Zada Y, Trnovski S, Burley E, Maestrado P. A comparison of continuous epidural-PCA analgesia for labor pain with or without maternal ambulation [abstract]. *Anesthesia & Analgesia* 2002;**94**(2S):Abstract no: S195.

#### COMET 2001 {published data only}

COMET Study Group. The comparative obstetric mobile epidural trial. Ambulatory epidural analgesia, delivery mode and pain relief: a randomised controlled trial [abstract]. *Anesthesiology* 2000;**92 Suppl**: Abstract no: A21.

COMET Study Group. The comparative obstetric mobile epidural trial (C.O.M.E.T.). Ambulatory epidural analgesia, delivery mode and pain relief: a randomized controlled trial. The C.O.M.E.T. Study Group. [abstract]. *European Journal of Anaesthesiology* 2000; **17**:782–3.

COMET Study Group, Wilson MJ. A randomised controlled trial comparing traditional with two "mobile" epidural techniques: effect on urinary catheterisation in labor [abstract]. *Anesthesiology* 2002;**96** (Suppl 1):Abstract no: Z2.

Comparative Obstetric Mobile Epidural Trial (COMET) Study Group UK. Effect of low-dose mobile versus traditional epidural tech-

Maternal positions and mobility during first stage labour (Review)

niques on mode of delivery: a randomised controlled trial. *Lancet* 2001;**358**(9275):19–23.

Duhig K, MacArthur C, Shennan AH, The COMET Study Group. The hypotensive and fetal heart rate response to low dose epidurals: analysis of an RCT data set [abstract]. *Journal of Obstetrics and Gynaecology* 2007;**27**(Suppl 1):S63–S64.

Elton C, Bharmal S, May AE, COMET Study Group. Does walking in labour with regional blockade affect the mode of delivery? [abstract]. *International Journal of Obstetric Anesthesia* 2002;**11 Suppl**: 33.

Hussain for the COMET Study Group. Haemodynamic changes with 'mobile' epidurals in labour: is it safe for women to ambulate? [abstract]. *Anesthesiology* 2001;**94**(1A):Abstract no: A63.

Shennan AH, COMET Study Group. The effect of low-dose 'mobile' compared with traditional epidural techniques on mode of delivery: a randomised controlled trial [abstract]. *Journal of Obstetrics and Gynaecology* 2001;**21 Suppl 1**:S19.

Wilson MJ, COMET Study Group. The comparative obstetric mobile epidural trial (C.O.M.E.T.). A randomized controlled trial [abstract]. *British Journal of Anaesthesia* 2001;**87**(4):659P.

Wilson MJ, Cooper G, MacArthur C, Shennan A, Comparative Obstetric Mobile Epidural Trial (COMET) Study Group UK. Randomized controlled trial comparing traditional with two "mobile" epidural techniques: anesthetic and analgesic efficacy. *Anesthesiology* 2002; **97**(6):1567–75.

#### Danilenko-Dixon 1996 {published data only}

Danilenko-Dixon DR, Tefft L, Cohen RA, Haydon B, Carpenter MW. Positional effects on maternal cardiac output during labor with epidural analgesia. *American Journal of Obstetrics and Gynecology* 1996;**175**:867–72.

#### Diaz 1980 {published data only}

Diaz AG, Schwarcz R, Fescina R, Caldeyro-Barcia R. Vertical position during the first stage of the course of labor, and neonatal outcome. *European Journal of Obstetrics & Gynecology and Reproductive Biology* 1980;**11**(1):1–7.

#### Divon 1985 {published data only}

Divon MY, Mushkat Y, Sarna Z, Zimmer EZ, Vilensky A, Paldi E. The association between fetal heart rate patterns and maternal posture during labor. *Archives of Gynecology* 1985;**237**(Suppl 1):89.

#### Ducloy-Bouthors 2006 {published data only}

Ducloy-Bouthors AS, De Gasquet B, Davette M, Cuisse M. Maternal postures and epidural analgesia during labour [Postures maternelles pendant le travail: description et interference avec l'analgesie peridurale]. *Annales Françaises d'Anesthesie et de Reanimation* 2006; **25**(6):605–8.

#### Hemminki 1983 {published data only}

Hemminki E, Saarikoski S. Ambulation and delayed amniotomy in the first stage of labor. *European Journal of Obstetrics & Gynecology* and Reproductive Biology 1983;**15**:129–39.

#### Hemminki 1985 {published data only}

Hemminki E, Lenck M, Saarikoski S, Henriksson L. Ambulation versus oxytocin in protracted labour: a pilot study. *European Journal of Obstetrics & Gynecology and Reproductive Biology* 1985;**20**:199–208.

#### Hillan 1984 {published data only}

Hillan EM. The birthing chair trial. Research and the Midwife Conference; 1984; Manchester, UK. 1984:22–37.

Stewart P, Hillan E, Calder A. A study of the benefits of maternal ambulation during labour and the use of a birth chair for delivery. Proceedings of 8th European Congress of Perinatal Medicine; 1982 Sept 7-10; Brussels, Belgium. 1982:Abstract no: 113.

Stewart P, Hillan E, Calder AA. A randomised trial to evaluate the use of a birth chair for delivery. Lancet 1983; Vol. 1:1296–8.

#### Hodnett 1982 {published data only}

Hodnett ED. Patient control during labor: effects of two types of fetal monitors. *Journal of Obstetric, Gynecologic and Neonatal Nursing* 1982;**2**:94–9.

#### Liu 1989 {published data only}

Liu, Y. The effects of the upright position during childbirth. *Image: Journal of Nursing Scholarship* 1989;**21**(1):14–8.

#### McCormick 2007 {published data only}

McCormick C. A randomised controlled trial of the effect of ambulation in the first stage of labour in terms of duration of labour of women with a previous caesarean section. controlled-trials.com (accessed 21 June 2007).

#### Melzack 1991 {published data only}

Melzack R, Belanger E, Lacroix R. Labor pain: effect of maternal position on front and back pain. *Journal of Pain and Symptom Management* 1991;**6**(8):476–80.

#### Molina 1997 {published data only}

Molina FJ, Sola PA, Lopez E, Pires C. Pain in the first stage of labor: relationship with the patient's position. *Journal of Pain and Symptom Management* 1997;**13**:98–103.

#### Radkey 1991 {published data only}

Radkey AL, Liston RM, Scott KE, Young C. Squatting: preventive medicine in childbirth?. Proceedings of Annual Meeting of Society of Obstetricians and Gynaecologists of Canada; 1991; Toronto, Ontario, Canada. 1991:76.

#### Read 1981 {published data only}

Read JA, Miller FC, Paul RH. Randomized trial of ambulation vs oxytocin for labor enhancement: a preliminary report. *American Journal of Obstetrics and Gynecology* 1981;**139**:669–72.

#### Schmidt 2001 {published data only}

Schmidt S, Sierra F, Hess C, Neubauer S, Kuhnert M, Heller G. Effect of modified labor posture on oxygenation of the fetus--a pulse oximetry study. *Zeitschrift fur Geburtshilfe und Neonatologie* 2001; **205**:49–53.

#### Schneider-Affeld 1982 {published data only}

Schneider-Affeld F, Martin K. Delivery from a sitting position. *Journal of Perinatal Medicine* 1982;2 Suppl:70–1.

#### Solano 1982 {published data only}

Solano F, Gallo M, Llamas C, Requena F, Arbues J. Perinatal effects of the maternal vertical position during dilatation period of the delivery. Effects on the mother-child couple [Efectos perinatales de la posición vertical materna durante el período de dilatación de parto. Efectos sobre el binomio madre–hijo]. *Acta Obstétrica y Ginecológica Hispano-Lusitana* 1982;**30**:81–104.

Maternal positions and mobility during first stage labour (Review)

#### Wu 2001 {published data only}

Wu X, Fan L, Wang Q. Correction of occipito-posterior by maternal postures during the process of labor. [Chinese]. *Chung-Hua Fu Chan Ko Tsa Chih [Chinese Journal of Obstetrics & Gynecology]* 2001;**36**(8): 468–9.

#### Additional references

#### Abitbol 1985

Abitbol MM. Supine position in labor and associated fetal heart rate changes. *Obstetrics and Gynecology* 1985;**65**(4):481–6.

#### Albers 1997

Albers LL, Anderson D, Cragin L, Daniels SM, Hunter C, Sedler KD, et al. The relationship of ambulation in labor to operative delivery. *Journal of Nurse Midwifery* 1997;**42**(1):4–8.

#### Boyle 2000

Boyle M. Childbirth in bed: the historical perspective. *Practising Midwife* 2000;**3**(11):21–4.

#### Caldeyro-Barcia 1960

Caldeyro-Barcia R, Noriega-Guerra L, Cibils LA, Alvarez H, Poseiro JJ, Pose SV, et al.Effect of position changes on the intensity and frequency of uterine contractions during labor. *American Journal of Obstetrics and Gynecology* 1960;**80**(2):284–90.

#### Carlson 1986

Carlson JM, Diehl JA, Sachtleben-Murray M, McRae M, Fenwick L, Friedman EA. Maternal position during parturition in normal labor. *Obstetrics & Gynecology* 1986;**68**(4):443–7.

#### Cluett 2002

Cluett ER, Nikodem VC, McCandlish RE, Burns EE. Immersion in water in pregnancy, labour and birth. *Cochrane Database of Systematic Reviews* 2002, Issue 2. [DOI: 10.1002/14651858.CD000111.pub2]

#### Cyna 2006

Cyna AM, Andrew M, Emmett RS, Middleton P, Simmons SW. Techniques for preventing hypotension during spinal anaesthesia for caesarean section. *Cochrane Database of Systematic Reviews* 2006, Issue 4. [DOI: 10.1002/14651858.CD002251.pub2]

#### Fenwick 1987

Fenwick L, Simkin P. Maternal positioning to prevent or alleviate dystocia in labor. *Clinical Obstetrics and Gynecology* 1987;**30**(1):83–9.

#### Gates 2005

Gates S. Methodological Guidelines. In: The Editorial Team. Pregnancy and Childbirth Group. About the Cochrane Collaboration (Collaborative Review Groups (CRGs)) 2005, Issue 2.

#### Goer 1999

Goer H. Does walking enhance labor progress?. *Birth* 1999;**26**(2): 127–9.

#### Gupta 2004

Gupta JK, Hofmeyr GJ, Smyth R. Position in the second stage of labour for women without epidural anaesthesia. *Cochrane Database of Systematic Reviews* 2004, Issue 1. [DOI: 10.1002/14651858.CD002006.pub2]

#### Higgins 2008

Higgins JPT, Green S, editors. Cochrane Handbook for Systematic Reviews of Interventions Version 5.0.0 [updated February 2008]. The Cochrane Collaboration, 2008. Available from www.cochrane-handbook.org.

### Hodnett 2007

Hodnett ED, Gates S, Hofmeyr GJ, Sakala C. Continuous support for women during childbirth. *Cochrane Database of Systematic Reviews* 2007, Issue 3. [DOI: 10.1002/14651858.CD003766.pub2]

#### Hunter 2007

Hunter S, Hofmeyr GJ, Kullier R. Hands and knees posture in late pregnancy or labour for fetal malpresention (lateral or posterior). *Cochrane Database of Systematic Reviews* 2007, Issue 3. [DOI: 10.1002/14651858.CD001063.pub3]

#### Huovinen 1979

Huovinen K, Teramo K. Effect of maternal position on fetal heart rate during extradural analgesia. *British Journal of Anaesthesia* 1979; **51**(8):767–73.

#### Lupe 1986

Lupe PJ, Gross TL. Maternal upright posture and mobility in labor: a review. *Obstetrics & Gynecology* 1986;**67**(5):727–34.

#### Marx 1982

Marx G. Hazards of the supine position in pregnancy. *Clinics in Obstetrics and Gynaecology* 1982;9(2):255–71.

#### Mendez-Bauer 1980

Mendez-Bauer C, Arroyo J, Freese U, Garcia-Ramos C, Hundsdorfer P, Izquierdo F, et al. The dynamics of labor in different positions. 7th European Congress Perinatal Medicine; 1980 Sept 14-17; Barcelona, Spain. 1980.

#### RevMan 2008

The Cochrane Collaboration. Review Manager (RevMan). 5.0. Copenhagen, The Nordic Cochrane Centre: The Cochrane Collaboration, 2008.

#### Roberts 1980

Roberts JE, Mendez-Bauer C. A perspective of maternal position during labor. *Journal of Perinatal Medcine* 1980;**8**:255–64.

#### Roberts 1983

Roberts JE, Mendez-Bauer C, Wodell DA. The effects of maternal position on uterine contractility and efficiency. *Birth* 1983;**10**(4): 243–9.

#### Roberts 1984

Roberts JE, Mendez-Bauer C, Blackwell J, Carpenter ME, Marchese T. Effects of lateral recumbency and sitting on the first stage of labor. *Journal of Reproductive Medicine* 1984;**29**(7):477–81.

#### Roberts 1989

Roberts J. Maternal position during the first stage of labour. In: Chalmers I, Enkin M, Keirse MJN editor(s). *Effective care in pregnancy and childbirth*. Vol. **2**, Oxford: Oxford University Press, 1989: 883–92.

#### Rooks 1999

Rooks JP. Evidence-based practice and its application to childbirth care for low-risk women. *Journal of Nurse Midwifery* 1999;44(4): 355–69.

#### Simkin 1989

Simkin P. Non-pharmacological methods of pain relief during labour. In: Chalmers I, Enkin M, Keirse MJN editor(s). *Effective care in* 

Maternal positions and mobility during first stage labour (Review)

pregnancy and chidbirth. Vol. 2, Oxford: Oxford University Press, 1989:893–911.

#### Simkin 2002

Simkin P, Hara M. Nonpharmacological relief of pain during labor: systematic reviews of five methods. *American Journal of Obstetrics and Gynecology* 2002;**186**(5):5131–59.

### Ueland 1969

Ueland K. Maternal cardiovascular dynamics. II posture and uterine contractions. *American Journal of Obstetrics and Gynecology* 1969; **103**:1–8.

#### Walsh 2000

Walsh D. Part five: why we should reject the 'bed birth' myth. *British Journal of Midwifery* 2000;**8**(9):554–8.

#### Williams 1980

Williams RM, Thom M, Studd J. A study of the benefits and acceptability of ambulation in spontaneous labour. *British Journal of Obstetrics and Gynaecology* 1980;**87**:122–6.

\* Indicates the major publication for the study

## CHARACTERISTICS OF STUDIES

## Characteristics of included studies [ordered by study ID]

#### Andrews 1990

Methods	Randomised trial.
Participants	40 women randomised. Inclusion criteria - nulliparous women with a medically uncomplicated pregnancy with a single vertex fetus in an anterior position, spontaneous onset of labour at 38 to 42 weeks' gestation, adequate pelvic measurements and intact amniotic membranes at the beginning of the maximum slope in their labour (4 to 9 cm dilatation).
Interventions	Intervention group: 20 - upright: standing, ambulating, sitting, squatting, or kneeling. Comparison group: 20 - recumbent: supine, lateral, or prone - hands and knees. All women - position assumed when cervical dilatation was from 4 to 9 cm; women were free to choose several variations within each position group. Women in both groups were free to assume positions from the other group for routines of care or rest; these activities were documented.
Outcomes	Length of first stage of labour. Pain. Analgesia amount.
Notes	Upright group - 15 women chose to lie down after receiving medication for rest; 5 of these women immediately returned to the upright position, stating that the contractions were more painful when they were lying down. The remaining 10 chose the lateral position to rest for up to 1 hour during the study period. Women in the recumbent position were monitored externally more often (n =13) than women in the upright position (n =1), which may have been an additional source of discomfort for women in the recumbent group.

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomly assigned'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.

#### Andrews 1990 (Continued)

Blinding? Outcome assessor	Unclear	Not stated.	
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.	
Bloom 1998			
Methods	Randomised trial.		
Participants	1067 women randomised. Inclusion criteria - women (nulliparous and multiparous) with uncomplicated pregnan- cies between 36 and 41 weeks' gestation and in active labour, having regular uterine con- tractions with cervical dilatation of 3 to 5 cm, and fetuses in cephalic presentation. Fetal membranes could be intact or ruptured. Exclusion criteria - women with any known complication of pregnancy, including breech presentations.		
Interventions	<ul> <li>complication of pregnancy, including breech presentations.</li> <li>Intervention group: 536 assigned to walking (walking as desired). Women were encouraged to walk but were instructed to return to their beds when they needed intravenous or epidural analgesia or when the second stage of labour began.</li> <li>Comparison group: 531 to labour in bed (usual care - confined to a labour bed). Women were permitted to assume their choice of supine, lateral or sitting positions during labour. All women - electronic fetal heart rate monitoring was not used routinely.</li> <li>Women whose fetuses had heart-rate abnormalities during routine surveillance conducted every 30 minutes with handheld Doppler devices, women who had meconium in the amniotic fluid, and women in whom labour was augmented by the administration of oxytocin underwent continuous electronic fetal monitoring, which prohibited further walking.</li> <li>Pelvic examinations were performed approximately every 3 hours - ineffective labour was suspected if the cervix did not dilate progressively during the first two hours after admission. If the fetal membranes were intact, amniotomy was performed. If a woman had hypotonic uterine contractions, and no further cervical dilatation after an additional 2-3 hours, labour was augmented by intravenous oxytocin (initial dose 6 mU per minute, increased every 40 mins by 6 mU per minute to a maximum of 42 mU per minute. Dystocia was diagnosed if labour had not progressed in 2-4 hours.</li> <li>In both study groups, the positions permitted during birth included the lateral (Sims') position and the dorsal-lithotomy position, with or without obstetrical stirrups.</li> <li>Women in both groups wore pedometers (for the walking group only, nurses recorded the number of minutes spent walking).</li> </ul>		

## Bloom 1998 (Continued)

Outcomes	Length of first stage labour. Length of second stage labour. Type of birth. Fetal distress. Analgesia.
	Fetal distress.
	Analgesia.
	Augmentation.
	Perineal trauma.
	Fetal distress.

## Notes

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomly assigned'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow-up.

## Broadhurst 1979

Methods	Randomised trial.
Participants	50 (8 primiparous and 17 multiparous in each group).
Interventions	Intervention group: 25 - ambulation. Comparison group: 25 - bed care.
Outcomes	Pain. Analgesia.
Notes	

## Broadhurst 1979 (Continued)

Risk of bias			
Item	Authors' judgement	Description	
Adequate sequence generation?	Unclear	Describled as 'randomly allocated'.	
Allocation concealment?	Unclear	Not stated.	
Blinding? Women	No	Not feasible.	
Blinding? Clinical staff	No	Not feasible.	
Blinding? Outcome assessor	Unclear	Not stated.	
Incomplete outcome data addressed? All outcomes	No	No losses to follow up.	

## Bundsen 1982

Methods	Randomised trial.		
Participants	60 women undergoing induction of labour.		
Interventions	Intervention group: 40 ambulation (telemetry). Comparison group: 20 bed care. All women: induced - primary amniotomy and immediate internal monitoring.		
Outcomes	Type of delivery. Pain.		
Notes			
Risk of bias			
Item	Authors' judgement	Description	

Adequate sequence generation?	Unclear	Described as 'randomization to three groups'.
Blinding? Women	No	

#### Bundsen 1982 (Continued)

Blinding? Clinical staff	No	
Blinding? Outcome assessor	No	
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up apparent.

## Calvert 1982

Methods	Quasi-randomised trial.	
Participants	200 women randomised. Inclusion criteria - women with a single fetus of at least 37 weeks' gestation; vertex presentation and no contraindication to vaginal birth; in spontaneous labour with uterine contractions occurring at least every 10 mins and a cervix at least 2.5 cm dilated. Exclusion criteria - women who had previously suffered a stillbirth or neonatal death or who had undergone a caesarean section.	
Interventions	Intervention: Ambulation with telemetry monitoring (women advised that they could get of bed to walk, sit in an easy chair or use the day room). Intervention group - ambulant women monitored with telemetry (n = 100). Comparison group - conventional cardiotocography (women nursed in bed) (n = 100). All women - all patients in bed were nursed in the lateral position or with a lateral tilt.	
Outcomes	Length of first stage. Type of delivery. Woman's pain. Analgesia. Length of second stage. Apgar < 7 at 5 mins.	
Notes	Telemetry group: 45% elected to get out of bed (and then only for short periods); average time out of bed = 1 hour 44 mins (range - 3 mins to 4 hours 20 mins) which was 30% of the monitored first stage of labour; 34 (75%) of those who left their beds initially elected to stay in bed by the time they reached a cervical dilatation of 7 cm.	
Risk of bias		
Item	Authors' judgement	Description
Adequate sequence generation?	No	Described as 'Final digit of hospital number (odd or even)'.

## Calvert 1982 (Continued)

Allocation concealment?	No	Described as 'Final digit of hospital number (odd or even)'.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.
Chan 1963		
Methods	Quasi-randomised tria	l.
Participants	200 women randomised. Inclusion criteria - primiparous. Exclusion criteria - planned elective caesarean section.	
Interventions	Intervention group:100 Comparison group: 10	) women were kept in the erect postion (sit or walk). 0 women were kept in a supine or lateral position.
Outcomes	Length of first stage. Type of delivery. Pain. Analgesia. Length of second stage. Fetal distress.	
Notes		

Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	No	Alternate group allocation.
Allocation concealment?	No	
Incomplete outcome data addressed? All outcomes	Unclear	No loss to follow up.

Chen 1987	
Methods	Quasi-randomised trial.
Participants	<ul> <li>116 women (185 women randomised, 116 included in the analyses).</li> <li>Inclusion criteria - women with uneventful pregnancies, full term, spontaneous labours, with a single fetus in cephalic presentation.</li> <li>Exclusion criteria - women received oxytocin augmentation; caesarean section due to cephalo-pelvic disproportion or fetal distress; women requested and received epidural anaesthesia; child with congenital anomalies; tococardiogram records were unsuitable for reading (n = 67 exclusions after group allocation).</li> </ul>
Interventions	Amniotomy performed when cervical dilatation reached 3 to 4 cm. Intervention group (sitting): Women free to assume any comfortable position in home- like part of obstetric unit (furnished with desk, chair, sofa but no bed). Most sat on a sofa (back of sofa at 65 degree angle from horizontal) with their knees flexed. When each woman's cervix became fully dilated, she was transferred to a birthing chair Comparison group (supine): Women to maintain dorsal or lateral recumbent position. No analgesia or anaesthesia used except for pudendal nerve block or perineal infiltration of xylocaine. Experimental group (1): sitting position during the entire course of labour (n = 41). Comparison groups (2): supine position in the first stage and birthing chair in the second stage (n = 32); (3): supine position throughout labour (n = 43).
Outcomes	Length of first stage. Type of delivery. Length of second stage. ARM.

## Notes

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	No	Described as 'Allocated following the order of their admission into the study'.
Allocation concealment?	No	Described as 'Allocated following the order of their admission into the study'.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.

## Chen 1987 (Continued)

Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	No	67 participants were excluded after group allocation (37%). Some of the reasons for exclusion are unlikely to have related to the intervention (e.g. children born with congenital abnomali- ties) but other reasons may have related to group allocation (e.g. oxytocin augmentation, caesarean for fetal distress).

## Collis 1999

Methods	Randomised trial
Participants	229 women (153 were in spontaneous labour and 76 had labour induced). Inclusion criteria - nulliparous women in spontaneous or induced labour who requested regional analgesia (given CSE); cephalic singleton pregnancy from 36 to 42 weeks' ges- tation, with no other pregnancy complications, e.g. pregnancy-induced hypertension.
Interventions	Intervention group: encouraged to spend at least 20 mins of each hour out of bed (n = $110$ ) - walking, standing, sitting in a rocking chair.
	either side.
	500-1000 ml Hartmann's solution infused as a preload; CSE - 27-G Becton-Dickinson Whitacre 119 mm spinal needle and 16-G Tuohy needle; long spinal needle inserted through Tuohy needle into cerebrospinal fluid (needle-through-needle CSE). Subarach- noid injection of 25 g fentanyl and 2.5 mg bupivacaine Labours were managed according to the department's standard practice (cervical dilata-
	tion was assessed every 3 hours and if dilatation had not increased by 2 cm, amniotomy was performed; if the membranes were intact, this was followed 2 hours later (if progress of labour was still inadequate) by augmentation of labour with oxytocin. If the membranes were ruptured and inadequate progress of labour was noted, then oxytocin was started without waiting for another 2 hours. The mothers were allowed up to 2 hours in the second stage of labour. If at the end of the second hour, birth was not imminent, instrumental delivery was performed.
Outcomes	Length of first stage. Type of delivery. Analgesia. Apgar.

## Collis 1999 (Continued)

Notes	51/110 women in the intervention group achieved at least 30% of time out of bed, 15 women spent no time out of bed, 44 spent 1 to 29%, 32 spent 30-59% and 19 women spent > 60% of time out of bed. Reasons for not ambulating: 16 women developed motor block, fatigue in 25 mothers, midwife instruction in 10 cases.
	Comparison group: 16/119 women got of bed (15 between 1-29% of the time and 1 between 30-59% of the time.
	Reasons for ambulating: passing urine.

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Not stated.
Allocation concealment?	Yes	Described as 'sealed opaque numbered envelopes'.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	Unclear	Described as 'Obstetrician was not aware which group the mother was in'.
Blinding? Outcome assessor	Unclear	Described as 'Obstetrician was not aware which group the mother was in'.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

#### Fernando 1994

Methods	Randomised trial.
Participants	40 nulliparous women receiving a CSE.
Interventions	Intervention group: out of bed (sitting in rocking chair, stand by bed, walk about) (n = 20). Comparison group: staying in bed (n = 20). All women - spinal injection of bupivacaine 2.5 mg and fentanyl 25 g using a 27 gauge, 1119 mm Becton-Dickinson Whitacre spinal needle through a 16 gauge Braun Tuohy needle, followed by epidural top ups of 10 mg bupivacaine in 10 ml with 2 g/ml of fentanyl.

## Fernando 1994 (Continued)

Outcomes	Apgar.	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomly allocated'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Flynn 1978

Methods	Randomised trial.	
Participants	68 (17 primigravidae and 17 multigravidae in each group, 33 cephalic and 1 breech presentation in each group).	
	Inclusion criteria - women in spontaneous labour.	
Interventions	<ul> <li>Intervention group: allowed to walk around while being continuously monitored by telemetry.</li> <li>When intravenous treatment was necessary (e.g. because of ketonuria or delay in labour) the women returned to bed.</li> <li>Comparison group: recumbent (nursed in the lateral position with conventional bedside monitoring of fetal heart and intrauterine pressure).</li> <li>All patients were nursed in bed during the second and third stages of labour.</li> <li>Dilatation of the cervix and station of the presenting part were assessed at the start of monitoring and every two to three hours during labour.</li> <li>Analgesia was administered when the midwife thought the woman was becoming distressed with pain.</li> </ul>	

## Flynn 1978 (Continued)

	Augmentation in labour with oxytocin or prostaglandin was given when indicated by delay in labour.
Outcomes	Length of first stage of labour. Type of delivery. Analgesia. Augmentation. Blood loss. Apgar.

## Notes

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomised prospective'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Frenea 2004

Methods	Randomised trial.
Participants	61 women. Inclusion criteria - women with uncomplicated term singleton pregnancies from 37 to 42 weeks' gestation in a fixed cephalic uncomplicated presentation, and 3 to 5 cm cervical dilatation at the time of epidural insertion. Women could be in spontaneous labour or admitted for elective induction. A normal fetal heart rate pattern was also required.
	Exclusion criteria - unfixed cephalic presentation, cervical dilatation more than 5 cm, a contraindication to epidural analgesia, or a systolic arterial blood pressure < 100 mmHg before epidural insertion, twin pregnancy, history of caesarean birth, and any known complications of pregnancy including breech presentation.

## Frenea 2004 (Continued)

Interventions	Intervention group: ambulation (n = 30). Women were asked to walk at least 15 mins of each hour or for 25% of the duration of the first stage of labour. Ambulation was permitted 15-20 mins after the initial injection, provided there was no postural hypotension, no motor block in lower limbs, no proprioception impairment and no fetal heart rate decelerations. The women were asked to return to bed when they requested an epidural top-up or if they experienced weakness or sensory changes. Walking ended when examination by a midwife revealed full cervical dilatation. Comparison group: recumbent (n = 31). Confined to bed in dorsal or lateral recumbent position. Monitoring of labour was as for the ambulatory group, but without telemetry. Epidural analgesia of intermittent administrations of 0.08% bupivacaine-epinephrine plus 1 g/ml of sufentanil.
Outcomes	Length of first stage of labour. Type of delivery. Analgesia.

## Notes

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Not stated.
Allocation concealment?	Yes	Described as 'sealed numbered envelopes'.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Haukkama 1982

Methods	Quasi-randomised trial.	
Participants	60 women. Inclusion criteria: healthy women with an uneventful pregnancy, giving birth between 38 and 42 weeks.	
Interventions	Intervention group: cardiotocography by telemetry (n = 31). Telemetry women were encouraged to sit or walk during the opening phase of labour. Comparison group: conventional cardiotocography (n = 29). All women - nitrous oxide-oxygen, pethidine (usual dose 75 mg given once or twice) or epidural block were used for analgesia when needed.	
Outcomes	Length of first stage. Type of delivery. Analgesia. Augmentation. ARM.	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as matched pairs 'allocated at random' to one of two

Thequite sequence generation.		groups.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Karraz 2003

Methods

Randomised trial.

## Karraz 2003 (Continued)

Participants	<ul> <li>221 (144 nulliparas - 97 (69.3%) in the ambulatory group and 47 (63.5%) in the non-ambulatory group.</li> <li>Inclusion criteria: women with uncomplicated singleton pregnancies who presented in spontaneous labour between 36 and 42 weeks' gestation or who were scheduled for induced labour.</li> <li>Study conducted in daytime only (as women in labour at night are less inclined to walk).</li> <li>Exclusion criteria - women with pre-eclampsia or previous caesarean.</li> </ul>		
Interventions	Intervention group: ambulatory (walked, sat in a chair or reclined in a semi-supine position (n = 141) - as long as they demonstrated: acceptable analgesia; acceptable systolic blood pressure and ability to stand on one leg. Comparison group: non-ambulatory (not allowed to sit, walk or go to the toilet); they had to remain in the supine position or to lie in a semi-supine or lateral position (n = 74). All - intermittent epidural injection of 0.1% ropivacaine with 0.6 $\mu$ g/ml sufentanil. Repeat injections were given when the women requested additional pain relief.		
Outcomes	Length of labour. Type of delivery. Analgesia. Augmentation.		
Notes			
Risk of bias			
Item	Authors' judgement	Description	
Adequate sequence generation?	Unclear	'Randomly divided' in a 2:1 ratio.	
Allocation concealment?	Unclear	Not stated.	
Blinding? Women	No	Not feasible.	
Blinding? Clinical staff	No	Not feasible.	
Blinding? Outcome assessor	Unclear	Not stated.	
Incomplete outcome data addressed? All outcomes	No	6 women were excluded after randomisation.	

MacLennan	1994
-----------	------

Methods	Randomised trial.
Participants	196 women. Inclusion criteria: women in spontaneous established labour (presence of regular con- tractions less than 10 mins apart and cervical dilatation of 3 cm or more) with a singleton fetus in a cephalic presentation between 37 and 42 weeks' gestation who had the ability to ambulate in labour. Exclusion criteria: women undergoing intravenous therapy, with hypertension (> 90 mmHg diastolic blood pressure), epidural or narcotic analgesia at or before entry to trial, evidence of possible fetal distress, previous prostaglandin treatment, induced labour and a physical inability to ambulate.
Interventions	Intervention group: ambulate as desired (n = 96). Women were encouraged to ambulate but were also given the option of sitting or lying down when they wished. Comparison group: recumbent. Most women chose a semi-recumbent posture with the head end of the bed at 45 degrees but they could also be on their side with lower elevation of the head. After entry to the trial, all women had an artificial rupture of the membranes if they had not already spontaneously ruptured.
Outcomes	Type of delivery. Analgesia. Augmentation. Apgar. Admission to NICU.
Notes	

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Yes	Described as 'Balanced variable blocks with stratification by par- ity'.
Allocation concealment?	Yes	Opaque, sealed envelopes.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.

## MacLennan 1994 (Continued)

Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.
McManus 1978		
Methods	Randomised trial.	
Participants	40 women (20 primigravidas and 20 having their second or third confinement). Inclusion criteria - gestational age 38 weeks or more, and cervical score 6 or greater. Exclusion criteria - multiple pregnancies or breech presentations.	
Interventions	Intervention group: upright - encouraged to "be up and about". If woman wished to go to bed, she was nursed in a sitting position with the aid of pillows. Comparsion group: recumbent - nursed in the lateral position Labour was induced by forewater amniotomy and 0.5 mg PGE2 immediately after amniotomy and hourly thereafter until labour was considered to be established. If labour was not established an hour after the 6th PGE2 tablet (i.e. 6 hours after amniotomy), intravenous oxytocin was given.	
Outcomes	Type of delivery. Analgesia. Augmentation. ARM. Blood loss. Fetal distress. Apgar.	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomised prospective study'.
Allocation concealment?	Unclear	Described as 'randomly allocated according to the contents of a

		plain envelope.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.

### McManus 1978 (Continued)

Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Miquelutti 2007

Methods	Randomised trial.
Participants	107 women attending a hospital in Brazil (2005-6). Inclusion criteria - low-risk nulliparous women, at term, in labour, aged 16 - 40 years. Cervical dilation between 3cm and 5cm. Singleton fetus in cephalic presentation. Exclusion criteria - contraindications to upright position or booked for elective caesarean section.
Interventions	Intervention group - (n = 54) women received written information/education involving the use of models on the benefits of maintaining an upright position and encouraged to stand, walk, sit, crouch or kneel. If women remained supine for more than 30 minutes they were encouraged to return to an upright position. Comparison group - (n = 53) routine care, women were not encouraged to adopt any position and were allowed to move around and adopt any position they chose.
Outcomes	Mode of delivery, duration of labour, augmentation, episiotomy, Apgar score, maternal preferences.
Notes	Women in the intervention group remained upright for 57% of the time compared to 28% for women in the comparison group.

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Yes	Computer generated.
Allocation concealment?	Yes	Sealed, opaque envelopes opened sequentially.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	
Blinding? Outcome assessor	No	

## Miquelutti 2007 (Continued)

Incomplete outcome data addressed? All outcomes	Yes	Few women lost to follow up.
Mitre 1974		
Methods	Randomised trial.	
Participants	100 women. Inclusion criteria - women who had been admitted to the labour room and had term pregnancies; were in the latent phase of labour or the active phase with the cervix between 1 and 3 cm; no medical stimulation of labour was required; no evidence of cephalopelvic disproportion; no history of surgery or trauma to the cervix; normal prenatal course; cephalic presentation.	
Interventions	Intervention group: sitting (n = 50). All women were allowed to sit up after the amniotomy had been performed and the presenting part was engaged. The women were allowed to lie down from time to time, if they desired. Comparison group: supine (n = 50). All women were placed in the supine position and allowed to turn on their sides. Direct fetal and maternal monitoring was performed randomly on several women in both groups, using a choriometric unit.	
Outcomes	Length first stage labour. Apgar.	
Notes		
Risk of bias		
Item	Authors' judgement	Description

Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'divided randomly into two groups'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.

## Mitre 1974 (Continued)

Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.
Nageotte 1997		
Methods	Randomised trial.	
Participants	761 (total of 3 arms; only 2 arms (n = 505) used here. Inclusion criteria: nulliparous women in spontaneous labour or with spontaneous rupture of membranes at 36 weeks or more with a fetus in the vertex position, who requested epidural analgesia.	
Interventions	Intervention group: ambulation encouraged (n = 253). Ambulation was defined as a minimum of five mins of walking per hour. Comparison group:ambulation discouraged (n = 252). All women had CSE. All women received a minimum of 1000 ml of lactated Ringer's solution intravenously during the 30 mins preceding the placement of the epidural needle. CSE - intrathecal narcotic with a continuous low-dose epidural infusion. After the location of the epidural space with an 18-gauge Tuohy needle, a 11.9 cm 27-gauge Whitacre spinal needle was passed through the epidural needle into the subarachnoid space. Then 10 g of sufentanil in 2 ml of normal saline was infused and the spinal needle removed. An epidural catheter was advanced 3 cm into the epidural space and a continuous infusion of 0.0625 % bupivacaine with 2 g of fentanyl per millilitre was given at a rate of 12 ml per hour. Subsequent bolus doses of epidural solution were given as requested (12 ml of 0.0625% bupivacaine).	
Outcomes	Type of delivery. Pain. Hypotension. Apgar.	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Adequate sequence generation?	Unclear	Described as 'randomly assigned'.
Allocation concealment?	Unclear	Not stated.
Blinding?	No	Not feasible.

Women

Maternal positions and mobility during first stage labour (Review) Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

## Nageotte 1997 (Continued)

Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

## Phumdoung 2007

Methods	Randomised trial. Randomised in blocks.
Participants	Women recruited from a hospital in Southern Thailand. (2 groups used in this analysis (n = 83)). Inclusion criteria - married, primiparous women aged 18 - 35 years and in latent phase for > 10 hours. Singleton fetus, cephalic presentation, gestation 38 - 42 weeks, fetal weight 2500 - 4000 g. Exclusion criteria - had analgesia before recruitment, induced labour, membrane rupture > 20 hours previously, psychiatric problem, infection, asthma or objection to intervention.
Interventions	5 separate intervention groups (described below). In this review we have included data from two groups: Intervention group - CAT position alternating half hourly with head high position (CAT position = facing towards bed head at 45 degrees with knees bent, taking weight on knees and elbows; head high position = lying at a 45-degree angle) (n = 40). Comparison group - supine in bed (n = 43).
Outcomes	Duration of first stage. Pain.
Notes	<ul> <li>Complicated study design with five study groups:</li> <li>1. CAT position alternating with head-high position with music (n = 40).</li> <li>2. CAT position alternating with head-high position (n = 40).</li> <li>3. CAT position alternating with supine position (n = 40).</li> <li>4. Head-high position (lying in bed on back at 45 degrees) (n = 41).</li> <li>5. Supine in bed (n = 43).</li> <li>In this review we have used data for groups 2 and 5 in the analyses.</li> <li>(It was not clear what 'CAT' signified)</li> </ul>
Risk of bias	

Item

Authors' judgement Description

## Phumdoung 2007 (Continued)

Adequate sequence generation?	Unclear	No information.
Allocation concealment?	Unclear	Described as 'random assignments'.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	
Blinding? Outcome assessor	No	
Incomplete outcome data addressed? All outcomes	Yes	Three women were lost to follow up as they had caesarean sec- tions during the first stage of labour. It was not clear whether this was before randomisation. No other loss to follow up was apparent.

## Vallejo 2001

Methods	Randomised trial.
Participants	160 women. Inclusion criteria - nulliparous women, 36 - 42 weeks' gestation, singleton pregnancy in the vertex position and 3-5 cm cervical dilatation at the time of epidural insertion. Women with uncomplicated pregnancies who presented in spontaneous labour or who were scheduled to be admitted for elective induction of labour (indications for elective IOL included post dates (> 42 weeks) and patient preference). Exclusion criteria - pre-eclampsia, diabetes mellitus, preterm gestation (< 36 weeks) and post-term gestation (> 42 weeks).
Interventions	Intervention group: AEA with ambulation, sitting in a chair or both (n = 75). After 1 hour, women with a modified Bromage score of 5 who could stand on one foot (right and left) without assistance (all women in this group were able to do this) and without hypotension (systolic blood pressure < 100 mmHg or a decrease of 20 mmHg), were encouraged to ambulate with a support person (spouse or friend). If the woman could not comply with ambulation, she was encouraged to sit in a chair. Ambulation was defined as a minimum of 5 min of walking per hour. Women were not allowed to ambulate if there were persistent fetal decelerations and were not allowed to be out of bed in the second stage of labour when women were actively pushing. Comparison group: AEA without ambulation or sitting in a chair (n = 76). Women were confined to bed, encouraged to stay recumbent in a lateral position, and were not allowed to raise the head of the bed more than 45 degrees from horizontal.

## Vallejo 2001 (Continued)

	All - AEA blocks initiated with 15-25 ml ropivacaine (0.07%) plus 100 g/ml fentanyl, no test dose, to achieve a T10 dermatome sensory level. After achieving adequate pain relief, a continuous infusion of 0.07% ropivacaine plus 2 g/ml fentanyl at 15-20 ml/hour was administered.
Outcomes	Length of first stage. Type of delivery. Pain. Length of second stage. Augmentation. Apgar.

Notes

Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	Yes	Described as 'random number computer-generated program'.
Allocation concealment?	Unclear	Not stated.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	No	9 women were excluded.

## Williams 1980

Methods	Quasi-randomised trial.
Participants	103 women (48 ambulant (25 primigravidae); 55 recumbent (30 primigravidae)). Inclusion criteria - women in spontaneous labour with no risk factors.
Interventions	Intervention group: ambulatory (n = 48). Women were informed about the possible benefits of ambulation and were encouraged to walk about during the first stage of labour

#### Williams 1980 (Continued)

	Women who refused ambulation or who requested to return to bed were allowed to do so. Any woman who developed abnormalities of the fetal heart rate or fresh meconium staining of the amniotic fluid was returned to bed Women who requested or who were advised to have an epidural also returned to bed but those requiring oxytocin augmentation of labour carried their intravenous infusions with them. Comparison group: recumbent (n = 55).
Outcomes	Length of first stage. Type of delivery. Analgesia. Length of second stage. Apgar score.

## Notes

## Risk of bias

Item	Authors' judgement	Description
Adequate sequence generation?	No	Described as 'divided into two groups according to their hospital number'.
Allocation concealment?	No	See above.
Blinding? Women	No	Not feasible.
Blinding? Clinical staff	No	Not feasible.
Blinding? Outcome assessor	Unclear	Not stated.
Incomplete outcome data addressed? All outcomes	Yes	No losses to follow up.

AEA: ambulatory epidural analgesia ARM: artificial rupture of the membranes CSE: combined spinal epidural G: gauge IOL: induction of labour Mins: minutes mU: mlli-units NICU: neonatal intensive care unit

Maternal positions and mobility during first stage labour (Review)

## Characteristics of excluded studies [ordered by study ID]

Ahmed 1985	Brief abstract, data for the single result presented were not in a form we were able use in the review.
Allahbadia 1992	Not clear that this was an RCT. States that 'patients were selected at random' but it was not clear that allocation to experimental and control groups was random.
Asselineau 1996	Not randomised.
Caldeyro-Barcia 1960	<ol> <li>Observational - Not RCT.</li> <li>Not all women were in the first stage of labour.</li> </ol>
Cobo 1968	Intervention not relevant. Study examining lying on side versus lying on back.
Cohen 2002	No outcomes relevant to the review reported.
COMET 2001	The trial compared low-dose combined spinal epidural and low-dose infusion techniques and traditional epidural techniques.
Danilenko-Dixon 1996	The purpose of this study was to compare cardiac output after epidural analgesia in both positions.
Diaz 1980	This study use quasi-randomised group allocation, but more than a third of the experimental group were excluded from the analysis; women that did not comply with the protocol were excluded post randomisation.
Divon 1985	No data relevant to the review were reported. Outcomes - BP, uterine work and beat to beat variability.
Ducloy-Bouthors 2006	Outcomes relevant to the review not reported
Hemminki 1983	In this study the comparison was between two management policies rather than two different treatments. One group was nursed in bed and one group was encouraged to mobilise but there were also other differences in the treatment the two groups received which may have had an effect on outcomes. Women nursed in bed had routine amniotomy, women in the ambulant group did not; monitoring was also different in the two groups. These differences in management mean that it is not possible to assess the effect of position on outcomes.
Hemminki 1985	Compared ambulation with immediate oxytocin.
Hillan 1984	Only randomised for the second stage of labour.
Hodnett 1982	All bed care patients had an epidural and not all ambulant patients did.
Liu 1989	Intervention not relevant, study deals with the second stage of labour.
McCormick 2007	Study not completed - no results reported.

## (Continued)

Melzack 1991	Cross-over design, no data reported for the first phase of the trial.
Molina 1997	Cross-over design, no data reported for the first phase of the trial.
Radkey 1991	Second stage of labour only.
Read 1981	Comparing ambulation with oxytocin.
Schmidt 2001	Does not fit primary objectives.
Schneider-Affeld 1982	No quantitative outcome data presented.
Solano 1982	Not randomised.
Wu 2001	Intervention not relevant to review outcomes. Study examining lying on one side rather than the other to correct fetal malpresentation.

BP: blood pressure

RCT: randomised controlled trial

## DATA AND ANALYSES

## Comparison 1. Upright and ambulant positions versus recumbent positions and bed care

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size	
1 Duration of first stage of labour	9	1677	Mean Difference (IV, Random, 95% CI)	-0.99 [-1.60, -0.39]	
(hours): all women					
1.1 Nulliparous women	8	927	Mean Difference (IV, Random, 95% CI)	-0.97 [-1.96, 0.02]	
1.2 Multiparous women	5	682	Mean Difference (IV, Random, 95% CI)	-0.52 [-1.04, 0.00]	
1.3 Mixed or unclear parity	1	68	Mean Difference (IV, Random, 95% CI)	-2.60 [-4.11, -1.09]	
2 Mode of birth: spontaneous	11	2217	Risk Ratio (M-H, Fixed, 95% CI)	1.01 [0.97, 1.05]	
vaginal; all women					
2.1 Nulliparous women	6	986	Risk Ratio (M-H, Fixed, 95% CI)	0.99 [0.93, 1.06]	
2.2 Multiparous women	4	647	Risk Ratio (M-H, Fixed, 95% CI)	1.02 [0.99, 1.06]	
2.3 Mixed or unclear parity	5	584	Risk Ratio (M-H, Fixed, 95% CI)	1.03 [0.94, 1.13]	
3 Mode of birth: operative/	10	2110	Risk Ratio (M-H, Fixed, 95% CI)	0.99 [0.78, 1.26]	
assisted: all women					
3.1 Nulliparous women	5	879	Risk Ratio (M-H, Fixed, 95% CI)	1.01 [0.72, 1.43]	
3.2 Multiparous women	4	647	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.24, 3.49]	
3.3 Mixed or unclear parity	5	584	Risk Ratio (M-H, Fixed, 95% CI)	0.98 [0.70, 1.39]	
4 Mode of birth: caesarean section:	10	2110	Risk Ratio (M-H, Fixed, 95% CI)	0.73 [0.51, 1.07]	
all women					
4.1 Nulliparous women	5	879	Risk Ratio (M-H, Fixed, 95% CI)	1.01 [0.61, 1.67]	
4.2 Multiparous women	4	647	Risk Ratio (M-H, Fixed, 95% CI)	0.39 [0.12, 1.24]	
4.3 Mixed or unclear parity	5	584	Risk Ratio (M-H, Fixed, 95% CI)	0.55 [0.29, 1.07]	
5 Maternal satisfaction	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
6 Maternal satisfaction	0	0	Mean Difference (IV, Fixed, 95% CI)	Not estimable	
7 Maternal pain	5		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only	
7.1 Complaints of discomfort/	2	230	Risk Ratio (M-H, Fixed, 95% CI)	1.21 [0.52, 2.81]	
labour more uncomfortable					
7.2 Requiring analgesia	4	1517	Risk Ratio (M-H, Fixed, 95% CI)	0.98 [0.93, 1.02]	
8 Maternal pain	1	40	Mean Difference (IV, Fixed, 95% CI)	0.74 [-0.27, 1.75]	
8.1 Comfort score	1	40	Mean Difference (IV, Fixed, 95% CI)	0.74 [-0.27, 1.75]	
9 Analgesia type	8		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only	
9.1 Opioid	7	1681	Risk Ratio (M-H, Fixed, 95% CI)	0.97 [0.89, 1.06]	
9.2 Epidural	8	1784	Risk Ratio (M-H, Fixed, 95% CI)	0.83 [0.72, 0.96]	
9.3 Entonox	3	300	Risk Ratio (M-H, Fixed, 95% CI)	1.04 [0.74, 1.47]	
10 Analgesia amount	1	40	Mean Difference (IV, Fixed, 95% CI)	-17.5 [-36.89, 1.89]	
10.1 Narcotics and other	1	40	Mean Difference (IV, Fixed, 95% CI)	-17.5 [-36.89, 1.89]	
analgesia					
11 Duration of second stage of	2	1170	Mean Difference (IV, Fixed, 95% CI)	1.22 [-1.32, 3.75]	
labour (minutes)					
11.1 Nulliparous women	2	599	Mean Difference (IV, Fixed, 95% CI)	5.04 [-2.45, 12.53]	
11.2 Multiparous women	2	571	Mean Difference (IV, Fixed, 95% CI)	0.72 [-1.97, 3.41]	
12 Augmentation of labour using	7	1540	Risk Ratio (M-H, Fixed, 95% CI)	0.90 [0.77, 1.06]	
oxytocin					
13 Artificial rupture of membranes	3	216	Risk Ratio (M-H, Fixed, 95% CI)	1.34 [0.59, 3.04]	

14 Hypotension requiring	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
15 Estimated blood loss > 500 ml	1	40	Risk Ratio (M-H, Fixed, 95% CI)	0.33 [0.01, 7.72]
16 Perineal trauma	1		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected
16.1 Episiotomy	1		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
16.2 Second-degree tears	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
16.3 Third-degree tears	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
16.4 Any tear	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
17 Fetal distress (requiring immediate delivery)	3	1307	Risk Ratio (M-H, Fixed, 95% CI)	0.54 [0.25, 1.18]
18 Use of neonatal mechanical ventilation	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
19 Apgar scores	6	679	Risk Ratio (M-H, Fixed, 95% CI)	0.70 [0.38, 1.28]
19.1 Apgar < 7 at 1 min	3	280	Risk Ratio (M-H, Fixed, 95% CI)	0.58 [0.30, 1.12]
19.2 Apgar < 7 at 5 mins	4	399	Risk Ratio (M-H, Fixed, 95% CI)	3.27 [0.34, 31.05]
19.3 apgar < 5 at birth	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
20 Admission to NICU	1	196	Risk Ratio (M-H, Fixed, 95% CI)	1.56 [0.45, 5.37]
21 Perinatal mortality	1	1067	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable

## Comparison 2. Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size	
1 Duration of first stage of labour: time of epidural insertion to complete cervical dilation (minutes)	3	433	Mean Difference (IV, Fixed, 95% CI)	14.14 [-15.23, 43.51]	
2 Mode of birth: spontaneous vaginal	5	1161	Risk Ratio (M-H, Fixed, 95% CI)	0.97 [0.89, 1.06]	
3 Mode of birth: operative spontaneous/assisted	5	1161	Risk Ratio (M-H, Fixed, 95% CI)	1.16 [0.93, 1.44]	
4 Mode of birth: caesarean section	5	1161	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.70, 1.19]	
5 Maternal satisfaction	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
6 Maternal pain	1	505	Risk Ratio (M-H, Fixed, 95% CI)	0.88 [0.70, 1.11]	
6.1 Requiring additional Bupivocaine bolus doses	1	505	Risk Ratio (M-H, Fixed, 95% CI)	0.88 [0.70, 1.11]	
7 Analgesia type	5		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected	
7.1 Opioid	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
7.2 Epidural	5		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	
8 Analgesia amount	4	843	Mean Difference (IV, Fixed, 95% CI)	-0.70 [-1.34, -0.06]	
8.1 Bupivocaine	3	463	Mean Difference (IV, Fixed, 95% CI)	-0.79 [-1.48, -0.09]	
8.2 Ropivacaine	1	151	Mean Difference (IV, Fixed, 95% CI)	19.70 [0.77, 38.63]	
8.3 Fentanyl	1	229	Mean Difference (IV, Fixed, 95% CI)	-0.38 [-1.99, 1.23]	
9 Duration of second stage of labour (minutes)	2	204	Mean Difference (IV, Fixed, 95% CI)	2.35 [-15.22, 19.91]	
10 Augmentation of labour using oxytocin	5	1161	Risk Ratio (M-H, Fixed, 95% CI)	0.98 [0.90, 1.07]	
11 Artificial rupture of membranes	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable	

12 Hypotension requiring intervention	3	781	Risk Ratio (M-H, Fixed, 95% CI)	1.12 [0.52, 2.45]
13 Estimated blood loss > 500 ml	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
14 Perineal trauma	0		Risk Ratio (M-H, Fixed, 95% CI)	Totals not selected
14.1 Episiotomy	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
14.2 Second-degree tears	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
14.3 Third-degree tears	0		Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
15 Fetal distress (requiring	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
immediate delivery)				
16 Use of neonatal mechanical ventilation	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
17 Apgar scores	5	986	Risk Ratio (M-H, Fixed, 95% CI)	1.02 [0.39, 2.66]
17.1 Apgar < 7 at 1 min	1	151	Risk Ratio (M-H, Fixed, 95% CI)	1.01 [0.31, 3.36]
17.2 Apgar < 7 at 5 mins	4	835	Risk Ratio (M-H, Fixed, 95% CI)	1.04 [0.21, 5.05]
18 Admission to NICU	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable
19 Perinatal mortality	0	0	Risk Ratio (M-H, Fixed, 95% CI)	Not estimable

## Comparison 3. Standing and walking versus non-ambulant positions

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Duration of the first stage of labour	8	927	Mean Difference (IV, Random, 95% CI)	-0.97 [-1.96, 0.02]
1.1 Studies where ambulation was encouraged	5	684	Mean Difference (IV, Random, 95% CI)	-0.20 [-1.36, 0.96]
1.2 Studies where women were not ambulant	3	243	Mean Difference (IV, Random, 95% CI)	-1.92 [-2.83, -1.01]

## Analysis I.I. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome I Duration of first stage of labour (hours): all women.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: I Duration of first stage of labour (hours): all women

	Oprigni	ł	Recumbent		Mean Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Random,95% CI		IV,Random,95% Cl
I Nulliparous women							
Haukkama 1982	13	10.2 (5.4)	12	8.9 (4.6)		2.0 %	1.30 [ -2.62, 5.22 ]
McManus 1978	10	10.5 (3.7)	10	10.5 (4.4)		2.3 %	0.0 [ -3.56, 3.56 ]
Williams 1980	25	7.9 (4.9)	30	7.4 (3.2)		4.5 %	0.50 [ -1.74, 2.74 ]
Chen 1987	22	3.25 (2.25)	38	4.23 (2.5)		8.0 %	-0.98 [ -2.21, 0.25 ]
Andrews 1990	20	3.9 (1.5)	20	5.41 (1.5)		9.3 %	-1.51 [ -2.44, -0.58 ]
Phumdoung 2007	40	3.54 (1.91)	43	6.33 (2.1)		9.6 %	-2.79 [ -3.65, -1.93 ]
Mitre 1974	50	5.47 (1.71)	50	7.25 (1.64)		10.4 %	-1.78 [ -2.44, -1.12 ]
Bloom 1998	272	7.6 (3.9)	272	7.3 (3.9)		10.4 %	0.30 [ -0.36, 0.96 ]
Subtotal (95% CI)	452		475		-	56.5 %	-0.97 [ -1.96, 0.02 ]
Heterogeneity: $Tau^2 = 1.38$ ;	$Chi^2 = 40$	.85, df = 7 (P<0.00	0001); I <sup>2</sup> =839	6			
Test for overall effect: $Z = 1$ .	.93 (P = 0.	054)					
2 Multiparous women							
Haukkama 1982	18	5.6 (3.8)	17	6.6 (4.1)		3.7 %	-1.00 [ -3.62, 1.62 ]
Williams 1980	23	6.3 (2.9)	25	7.8 (5.4)		4.1 %	-1.50 [ -3.93, 0.93 ]
McManus 1978	10	5.3 (1.4)	10	5.6 (2.1)		6.6 %	-0.30 [ -1.86, 1.26 ]
Chen 1987	19	1.2 (0.75)	37	2.08 (1.08)		11.0 %	-0.88 [ -1.36, -0.40 ]
Bloom 1998	264	4.6 (2.4)	259	4.7 (2.4)	-	11.3 %	-0.10 [ -0.51, 0.31 ]
Subtotal (95% CI)	334		348		•	36.7 %	-0.52 [ -1.04, 0.00 ]
Heterogeneity: $Tau^2 = 0.12$ ;	$Chi^2 = 6.7$	'3, df = 4 (P = 0.15	5);   <sup>2</sup> =4 %				
Test for overall effect: $Z = 1$ .	.95 (P = 0.	051)					
3 Mixed or unclear parity							
Flynn 1978	34	4.1 (3.17)	34	6.7 (3.17)	• <b>B</b>	6.8 %	-2.60 [ -4.11, -1.09 ]
Subtotal (95% CI)	34		34	-		6.8 %	-2.60 [ -4.11, -1.09 ]
Test for overall effect: $7 = 3$	28 (P - 0)	00072)					
Total (95% CI)	820	00072)	857		•	100.0 %	-0.99 [ -1.60, -0.39 ]
Heterogeneity: $Tau^2 = 0.80;$	Chi <sup>2</sup> = 62	.16, df = 13 (P<0.0	00001); I <sup>2</sup> =79	9%			
Test for overall effect: $Z = 3$ .	.23 (P = 0.						

-2 0

Favours upright Favours recumbent

## Analysis 1.2. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 2 Mode of birth: spontaneous vaginal; all women.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 2 Mode of birth: spontaneous vaginal; all women

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Nulliparous women	F/10	F/10		
Inicimanus 1978	5/10	5/10		1.00 [ 0.42, 2.40 ]
Williams 1980	17/25	21/30	—	0.97 [ 0.68, 1.39 ]
Chen 1987	22/22	30/38	+	1.25 [ 1.05, 1.49 ]
Miquelutti 2007	31/54	36/53	-+-	0.85 [ 0.63, 1.14 ]
Chan 1963	71/100	70/100	+	1.01 [ 0.85, 1.21 ]
Bloom 1998	232/272	236/272	•	0.98 [ 0.92, 1.05 ]
Subtotal (95% CI)	483	503	•	0.99 [ 0.93, 1.06 ]
Heterogeneity: $Chi^2 = 7.94$ , df = Test for overall effect: Z = 0.28 ( 2 Multiparous women	(Recumbent) : 5 (P = 0.16); l <sup>2</sup> =37% (P = 0.78)			
Williams 1980	23/23	25/25		0.0 [ 0.0, 0.0 ]
McManus 1978	10/10	9/10	+-	1.11 [ 0.85, 1.44 ]
Chen 1987	18/19	36/37	+	0.97 [ 0.86, 1.10 ]
Bloom 1998	258/264	247/259	•	1.02 [ 0.99, 1.06 ]
Subtotal (95% CI)	316	331	•	1.02 [ 0.99, 1.06 ]
Total events: 309 (Upright), 317 Heterogeneity: $Chi^2 = 1.00$ , df = Test for overall effect: Z = 1.41 ( 3 Mixed or unclear parity	(Recumbent) 2 (P = 0.61); l <sup>2</sup> =0.0% P = 0.16)			
Bundsen 1982	38/40	17/20		1.12 [ 0.92, 1.36 ]
Flynn 1978	32/34	23/34		1.39 [ 1.09, 1.78 ]
Haukkama 1982	27/31	24/29	+	1.05 [ 0.85, 1.30 ]
MacLennan 1994	64/96	72/100	+	0.93 [ 0.77, 1.12 ]
Calvert 1982	77/100	78/100	+	0.99 [ 0.85, 1.15 ]
Subtotal (95% CI)	301	283	•	1.03 [ 0.94, 1.13 ]
Total events: 238 (Upright), 214 Heterogeneity: $Chi^2 = 7.92$ , df =	(Recumbent) = 4 (P = 0.09); I <sup>2</sup> =50%			
			0.1 0.2 0.5 2 5 10 Favours recumbent Favours upright	

(Continued . . . )

Maternal positions and mobility during first stage labour (Review)

Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

45

					( Continued)	
Study or subgroup	Upright	Recumbent	F	Risk Ratio	Risk Ratio	
	n/N	n/N	M-H,Fi>	(ed,95% Cl	M-H,Fixed,95% Cl	
Test for overall effect: Z = 0.67	(P = 0.50)					
Total (95% CI)	1100	1117		•	1.01 [ 0.97, 1.05 ]	
Total events: 925 (Upright), 929	9 (Recumbent)					
Heterogeneity: Chi <sup>2</sup> = 17.67, d	$f = 13 (P = 0.17); I^2 = 26\%$					
Test for overall effect: $Z = 0.59$	(P = 0.56)					
			0.1 0.2 0.5	2 5 10		
			Favours recumbent	Favours upright		

## Analysis I.3. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 3 Mode of birth: operative/assisted: all women.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 3 Mode of birth: operative/assisted: all women

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Nulliparous women				
McManus 1978	4/10	4/10	—	1.00 [ 0.34, 2.93 ]
Chen 1987	0/22	8/38		0.10 [ 0.01, 1.65 ]
Williams 1980	7/25	8/30	-	1.05 [ 0.44, 2.49 ]
Bloom 1998	21/272	15/272	-	1.40 [ 0.74, 2.66 ]
Chan 1963	21/100	21/100	+	1.00 [ 0.58, 1.71 ]
Subtotal (95% CI)	429	450	+	1.01 [ 0.72, 1.43 ]
Total events: 53 (Upright), 56 (Re	ecumbent)			
Heterogeneity: $Chi^2 = 3.6I$ , df =	$= 4 (P = 0.46); I^2 = 0.0\%$			
Test for overall effect: $Z = 0.07$ (	P = 0.94)			
2 Multiparous women				
Williams 1980	0/23	0/25		0.0 [ 0.0, 0.0 ]
Chen 1987	1/19	1/37		1.95 [ 0.13, 29.45 ]
McManus 1978	0/10	1/10		0.33 [ 0.02, 7.32 ]
Bloom 1998	2/264	2/259		0.98 [ 0.14, 6.91 ]
Subtotal (95% CI)	316	331	-	0.91 [ 0.24, 3.49 ]
			<u> </u>	
			0.01 0.1 1 10 100	

0.01 0.1 1 10 100 Favours upright Favours recumbent

(Continued . . . )

Maternal positions and mobility during first stage labour (Review)

Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

46

Study or subgroup	Upright	Recumbent	Risk Ratio	( Continued) Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
Total events: 3 (Upright), 4 (Recu	umbent)			
Heterogeneity: $Chi^2 = 0.71$ , df =	2 (P = 0.70); I <sup>2</sup> =0.0%			
Test for overall effect: $Z = 0.14$ (	P = 0.89)			
3 Mixed or unclear parity				
Haukkama 1982	4/31	3/29		1.25 [ 0.30, 5.10 ]
Bundsen 1982	2/40	3/20		0.33 [ 0.06, 1.84 ]
Flynn 1978	2/34	10/34		0.20 [ 0.05, 0.85 ]
Calvert 1982	18/100	15/100	-	1.20 [ 0.64, 2.25 ]
MacLennan 1994	26/96	21/100	+	1.29 [ 0.78, 2.13 ]
Subtotal (95% CI)	301	283	+	0.98 [ 0.70, 1.39 ]
Total events: 52 (Upright), 52 (Re	ecumbent)			
Heterogeneity: $Chi^2 = 7.85$ , df =	4 (P = 0.10); l <sup>2</sup> =49%			
Test for overall effect: $Z = 0.10$ (	P = 0.92)			
Total (95% CI)	1046	1064	<b>†</b>	0.99 [ 0.78, 1.26 ]
Total events: 108 (Upright), 112	(Recumbent)			
Heterogeneity: $Chi^2 = 12.21$ , df	= 12 (P = 0.43); 1 <sup>2</sup> =2%			
Test for overall effect: $Z = 0.05$ (	P = 0.96)			
			0.01 0.1 1 10 100	
			Favours upright Favours recurr	bent

## Analysis I.4. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 4 Mode of birth: caesarean section: all women.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 4 Mode of birth: caesarean section: all women

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
I Nulliparous women				
Chen 1987	0/22	0/38		0.0 [ 0.0, 0.0 ]
Williams 1980	1/25	1/30		1.20 [ 0.08, 18.23 ]
McManus 1978	1/10	1/10		1.00 [ 0.07, 13.87 ]
Chan 1963	7/100	5/100		1.40 [ 0.46, 4.26 ]
Bloom 1998	19/272	21/272	+	0.90 [ 0.50, 1.64 ]
Subtotal (95% CI)	429	450	+	1.01 [ 0.61, 1.67 ]
Total events: 28 (Upright), 28 (Re	ecumbent)			
Heterogeneity: $Chi^2 = 0.48$ , df =	= 3 (P = 0.92); I <sup>2</sup> =0.0%			
Test for overall effect: $Z = 0.03$ (	(P = 0.98)			
2 Multiparous women				
McManus 1978	0/10	0/10		0.0 [ 0.0, 0.0 ]
Williams 1980	0/23	0/25		0.0 [ 0.0, 0.0 ]
Chen 1987	0/19	0/37		0.0 [ 0.0, 0.0 ]
Bloom 1998	4/264	10/259		0.39 [ 0.12, 1.24 ]
Subtotal (95% CI)	316	331	-	0.39 [ 0.12, 1.24 ]
Total events: 4 (Upright), 10 (Ree	cumbent)			
Heterogeneity: $Chi^2 = 0.0$ , df =	0 (P = 1.00); $I^2 = 0.0\%$			
Test for overall effect: $Z = 1.60$ (	(P = 0.11)			
3 Mixed or unclear parity				
Flynn 1978	0/34	1/34		0.33 [ 0.01, 7.91 ]
Haukkama 1982	0/31	2/29		0.19 [ 0.01, 3.75 ]
Bundsen 1982	0/40	3/20	· • •	0.07 [ 0.00, 1.35 ]
MacLennan 1994	6/96	7/100		0.89 [ 0.31, 2.56 ]
Calvert 1982	5/100	7/100		0.71 [ 0.23, 2.18 ]
Subtotal (95% CI)	301	283	•	0.55 [ 0.29, 1.07 ]
Total events: 11 (Upright), 20 (Re	ecumbent)			
Heterogeneity: $Chi^2 = 3.45$ , df =	= 4 (P = 0.49); l <sup>2</sup> =0.0%			
Test for overall effect: $Z = 1.77$ (	(P = 0.077)			
			0.01 0.1 1 10 100	
			Favours upright Favours recumbent	

(Continued ...)



## Analysis 1.7. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 7 Maternal pain.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 7 Maternal pain

Study or subgroup	Upright	Recumbent	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl		M-H,Fixed,95% Cl
I Complaints of discomfort/lal	bour more uncomfc	ortable			
Chan 1963	8/100	0/100		5.9 %	17.00 [ 0.99, 290.62 ]
Haukkama 1982	2/17	7/13	← <b></b>	94.1 %	0.22 [ 0.05, 0.88 ]
Subtotal (95% CI)	117	113	-	100.0 %	1.21 [ 0.52, 2.81 ]
Total events: 10 (Upright), 7 (F	Recumbent)				
Heterogeneity: Chi <sup>2</sup> = 9.12, d	$f =   (P = 0.003);  ^2$	=89%			
Test for overall effect: $Z = 0.45$	5 (P = 0.65)				
2 Requiring analgesia					
Broadhurst 1979	9/25	25/25		4.1 %	0.37 [ 0.22, 0.62 ]
Chan 1963	45/100	35/100		5.7 %	1.29 [ 0.91, 1.81 ]
Calvert 1982	98/100	98/100	•	15.9 %	1.00 [ 0.96, 1.04 ]
Bloom 1998	452/536	455/531	•	74.3 %	0.98 [ 0.94, 1.04 ]
Subtotal (95% CI)	761	756	•	100.0 %	0.98 [ 0.93, 1.02 ]
Total events: 604 (Upright), 61	3 (Recumbent)				
Heterogeneity: Chi <sup>2</sup> = 17.43, o	df = 3 (P = 0.00058	); l <sup>2</sup> =83%			
Test for overall effect: $Z = 0.92$	3 (P = 0.35)				
			0.1 0.2 0.5 2 5 10		
			Favours upright Favours recumbent		

## Analysis I.8. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 8 Maternal pain.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 8 Maternal pain

Study or subgroup	Upright		Recumbent		Mei	an Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fixe	ed,95% Cl		IV,Fixed,95% CI
I Comfort score								
Andrews 1990	20	12.53 (1.63)	20	11.79 (1.63)			100.0 %	0.74 [ -0.27, 1.75 ]
Total (95% CI)	20		20			•	100.0 %	0.74 [ -0.27, 1.75 ]
Heterogeneity: not ap	plicable							
Test for overall effect:	Z = 1.44 (P =	= 0.15)						
					-10 -5	0 5 10		
					Favours upright	Favours recun	nbent	

## Analysis I.9. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 9 Analgesia type.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 9 Analgesia type

Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
l Opioid					
McManus 1978	19/20	16/20		3.4 %	1.19 [ 0.93, 1.51 ]
Broadhurst 1979	9/25	18/25	<u> </u>	3.9 %	0.50 [ 0.28, 0.89 ]
Haukkama 1982	16/31	21/29		4.7 %	0.71 [ 0.47, 1.07 ]
Flynn 1978	14/34	26/34		5.6 %	0.54 [ 0.35, 0.84 ]
MacLennan 1994	39/96	40/100	+	8.4 %	1.02 [ 0.72, 1.43 ]
Calvert 1982	73/100	73/100	+	15.7 %	1.00 [ 0.84, 1.18 ]
Bloom 1998	285/536	271/531	=	58.4 %	1.04 [ 0.93, 1.17 ]
Subtotal (95% CI) Total events: 455 (Upright), 46	<b>842</b> 55 (Recumbent)	839	•	100.0 %	0.97 [ 0.89, 1.06 ]

Favours upright Favours recumbent

(Continued . . . )

Maternal positions and mobility during first stage labour (Review)

Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

50

Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	( Continued) Risk Ratio M-H,Fixed,95% Cl
Heterogeneity: Chi <sup>2</sup> = 18.32,	$df = 6 (P = 0.01); I^2$	=67%			
Test for overall effect: $Z = 0.6$	I (P = 0.54)				
2 Epidural					
McManus 1978	4/20	2/20		0.7 %	2.00 [ 0.41, 9.71 ]
Williams 1980	3/48	4/55		1.3 %	0.86 [ 0.20, 3.65 ]
Haukkama 1982	3/31	5/29		1.8 %	0.56 [ 0.15, 2.14 ]
Broadhurst 1979	0/25	7/25	·	2.6 %	0.07 [ 0.00,  .   ]
Flynn 1978	0/34	8/34	·	3.0 %	0.06 [ 0.00, 0.98 ]
Calvert 1982	16/100	22/100		7.7 %	0.73 [ 0.41, 1.30 ]
MacLennan 1994	43/96	52/100		17.9 %	0.86 [ 0.64, 1.15 ]
Bloom 1998	167/536	184/531	-	64.9 %	0.90 [ 0.76, 1.07 ]
Subtotal (95% CI)	890	894	•	100.0 %	0.83 [ 0.72, 0.96 ]
Total events: 236 (Upright), 28	84 (Recumbent)				
Heterogeneity: $Chi^2 = 9.04$ , d	$f = 7 (P = 0.25); I^2$	=23%			
Test for overall effect: $Z = 2.5$	3 (P = 0.011)				
3 Entonox					
McManus 1978	1/20	2/20		5.4 %	0.50 [ 0.05, 5.08 ]
Calvert 1982	18/100	13/100		35.4 %	1.38 [ 0.72, 2.67 ]
Haukkama 1982	20/31	21/29	-	59.1 %	0.89 [ 0.63, 1.26 ]
Subtotal (95% CI)	151	149	+	100.0 %	1.04 [ 0.74, 1.47 ]
Total events: 39 (Upright), 36	(Recumbent)				
Heterogeneity: $Chi^2 = 1.91$ , d	$f = 2 (P = 0.38); I^2$	=0.0%			
Test for overall effect: $Z = 0.2$	5 (P = 0.80)				
			0.1 0.2 0.5 [ 2 5 10		

Favours upright Favours recumbent

## Analysis 1.10. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 10 Analgesia amount.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 10 Analgesia amount

Study or subgroup	Upright		Recumbent			Mea	n Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fixe	d,95% Cl		IV,Fixed,95% CI
I Narcotics and othe	r analgesia								
Andrews 1990	20	21.25 (31.29)	20	38.75 (31.29)		-		100.0 %	-17.50 [ -36.89, 1.89 ]
Total (95% CI)	20		20			•		100.0 %	-17.50 [ -36.89, 1.89 ]
Heterogeneity: not ap	plicable								
Test for overall effect:	Z = 1.77 (F	P = 0.077)							
					-100	-50 (	50 10	0	
					Favours	upright	Favours recur	mbent	

## Analysis I.II. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome II Duration of second stage of labour (minutes).

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: II Duration of second stage of labour (minutes)

Study or subgroup	Upright		Recumbent		Me	an Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fi>	ed,95% Cl		IV,Fixed,95% CI
I Nulliparous women								
Williams 1980	25	45 (34)	30	43 (21)	-		2.7 %	2.00 [ -13.30, 17.30 ]
Bloom 1998	272	60 (54)	272	54 (48)			8.7 %	6.00 [ -2.59,  4.59 ]
Subtotal (95% CI)	297		302			•	11.4 %	5.04 [ -2.45, 12.53 ]
Heterogeneity: $Chi^2 = 0.2$	0, df = 1 (P =	= 0.65); I <sup>2</sup> =0.0%						
Test for overall effect: $Z =$	I.32 (P = 0.	19)						
2 Multiparous women								
Williams 1980	23	17 (11)	25	14 (8)		-	21.3 %	3.00 [ -2.48, 8.48 ]
Bloom 1998	264	12 (18)	259	12 (18)		<b>=</b>	67.3 %	0.0 [ -3.09, 3.09 ]
Subtotal (95% CI)	287		284			•	88.6 %	0.72 [ -1.97, 3.41 ]
Heterogeneity: Chi <sup>2</sup> = 0.8	7, df = 1 (P =	= 0.35); I <sup>2</sup> =0.0%						
					-50 -25	0 25 50	1	
				F	avours upright	Favours recur	nbent	
								(Continued )

Maternal positions and mobility during first stage labour (Review)



## Analysis 1.12. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 12 Augmentation of labour using oxytocin.

Comparison: I Upright an	d ambulant positions versu	is recumbent positions and bed	care	
Outcome: 12 Augmentation	on of labour using oxytocir	1		
Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Risk Ratio M-H,Fixed,95% Cl
Chen 1987	0/1	0/1		0.0 [ 0.0, 0.0 ]
McManus 1978	2/20	4/20		0.50 [ 0.10, 2.43 ]
MacLennan 1994	15/96	2/ 00		1.30 [ 0.64, 2.64 ]
Flynn 1978	6/34	12/34		0.50 [ 0.21, 1.18 ]
Haukkama 1982	20/31	19/29		0.98 [ 0.68, 1.43 ]
Miquelutti 2007	25/54	24/53		1.02 [ 0.68, 1.54 ]
Bloom 1998	122/536	37/53	-	0.88 [ 0.71, 1.09 ]
Total (95% CI) Total events: 190 (Upright), 2 Heterogeneity: $Chi^2 = 4.01$ , 0 Test for overall effect: $Z = 1$ .	<b>772</b> 208 (Recumbent) df = 5 (P = 0.55); I <sup>2</sup> =0.0% 22 (P = 0.22)	<b>768</b>	•	0.90 [ 0.77, 1.06 ]
			0.1 0.2 0.5 1 2 5 10 Favours upright Favours recumbent	

Maternal positions and mobility during first stage labour (Review)

Review: Maternal positions and mobility during first stage labour

-

## Analysis 1.13. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 13 Artificial rupture of membranes.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 13 Artificial rupture of membranes

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
Chen 1987	41/41	75/75		0.0 [ 0.0, 0.0 ]
McManus 1978	20/20	20/20		0.0 [ 0.0, 0.0 ]
Haukkama 1982	10/31	7/29		1.34 [ 0.59, 3.04 ]
Total (95% CI)	92	124	-	1.34 [ 0.59, 3.04 ]
Total events: 71 (Upright), 10	02 (Recumbent)			
Heterogeneity: $Chi^2 = 0.0$ , d	$f = 0 (P = 1.00); I^2 = 0.0\%$			
Test for overall effect: $Z = 0$ .	69 (P = 0.49)			
			0.1 0.2 0.5 1 2 5 10	
			Favours upright Favours recumbent	

## Analysis 1.15. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 15 Estimated blood loss > 500 ml.

Review: Maternal positi	ions and mobility dur	ing first stage labour				
Comparison: I Upright	and ambulant positi	ons versus recumbent	positions and bed car	e		
Outcome: 15 Estimated	d blood loss > 500 n	h				
Study or subgroup	Upright n/N	Recumbent n/N	F M-H,Fi>	Risk Ratio ked,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
McManus 1978	0/20	1/20			100.0 %	0.33 [ 0.01, 7.72 ]
<b>Total (95% CI)</b> Total events: 0 (Upright), Heterogeneity: not applic: Test for overall effect: Z =	<b>20</b> I (Recumbent) able = 0.69 (P = 0.49)	20			100.0 %	0.33 [ 0.01, 7.72 ]
			0.01 0.1 Favours upright	I 10 100 Favours recumbent		

Maternal positions and mobility during first stage labour (Review)

## Analysis 1.16. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 16 Perineal trauma.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 16 Perineal trauma

Study or subgroup	Upright	Recumbent	F	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,H>	(ed,95% Cl	M-H,Fixed,95% CI
I Episiotomy					
Bloom 1998	122/536	124/531	-	-	0.97 [ 0.78, 1.21 ]
2 Second-degree tears					
3 Third-degree tears					
4 Any tear					
			<b>i</b> 1	, <u>, , , , , , , , , , , , , , , , , , </u>	
			0.01 0.1	1 10 100	
			Favours upright	Favours recumbent	

## Analysis 1.17. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 17 Fetal distress (requiring immediate delivery).

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 17 Fetal distress (requiring immediate delivery)

Study or subgroup	Upright	Recumbent	1	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fi	xed,95% Cl		M-H,Fixed,95% Cl
McManus 1978	0/20	1/20			8.5 %	0.33 [ 0.01, 7.72 ]
Chan 1963	4/100	4/100		•	22.8 %	1.00 [ 0.26, 3.89 ]
Bloom 1998	5/536	12/531		_	68.7 %	0.41 [ 0.15, 1.16 ]
Total (95% CI)	656	651	•		100.0 %	0.54 [ 0.25, 1.18 ]
Total events: 9 (Upright),	17 (Recumbent)					
Heterogeneity: Chi <sup>2</sup> = 1.1	4, df = 2 (P = 0.57);	l <sup>2</sup> =0.0%				
Test for overall effect: Z =	= 1.55 (P = 0.12)					
			0.01 0.1	1 10 100		
			Favours upright	Favours recumbent		

Maternal positions and mobility during first stage labour (Review)

## Analysis 1.19. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 19 Apgar scores.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 19 Apgar scores

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
Apgar < 7 at   min				
Fernando 1994	2/20	2/20	<b>_</b>	1.00 [ 0.16, 6.42 ]
McManus 1978	0/20	2/20		0.20 [ 0.01, 3.92 ]
Calvert 1982	10/100	17/100	-	0.59 [ 0.28, 1.22 ]
Subtotal (95% CI)	140	140	•	0.58 [ 0.30, 1.12 ]
Total events: 12 (Upright), 21 (R	ecumbent)			
Heterogeneity: $Chi^2 = 0.82$ , df =	= 2 (P = 0.66); I <sup>2</sup> =0.0%			
Test for overall effect: $Z = 1.62$ (	(P = 0.11)			
2 Apgar < 7 at 5 mins				
Haukkama 1982	0/31	0/29		0.0 [ 0.0, 0.0 ]
Fernando 1994	0/20	0/20		0.0 [ 0.0, 0.0 ]
Williams 1980	1/48	0/55		3.43 [ 0.14, 82.25 ]
MacLennan 1994	1/96	0/100		3.12 [ 0.13, 75.75 ]
Subtotal (95% CI)	195	204		3.27 [ 0.34, 31.05 ]
Total events: 2 (Upright), 0 (Recu	umbent)			
Heterogeneity: $Chi^2 = 0.00$ , df =	= I (P = 0.97); I <sup>2</sup> =0.0%			
Test for overall effect: $Z = 1.03$ (	(P = 0.30)			
3 apgar < 5 at birth				
Subtotal (95% CI)	0	0		0.0 [ 0.0, 0.0 ]
Total events: 0 (Upright), 0 (Recu	umbent)			
Heterogeneity: not applicable				
Test for overall effect: not applica	able			
Total (95% CI)	335	344	•	0.70 [ 0.38, 1.28 ]
Total events: 14 (Upright), 21 (R	ecumbent)			
Heterogeneity: $Chi^2 = 2.84$ , df =	= 4 (P = 0.58); l <sup>2</sup> =0.0%			
Test for overall effect: $Z = 1.16$ (	(P = 0.24)			

0.01 0.1 10 100 Favours upright Favours recumbent

Maternal positions and mobility during first stage labour (Review)

## Analysis 1.20. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 20 Admission to NICU.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 20 Admission to NICU

Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
MacLennan 1994	6/96	4/100		100.0 %	1.56 [ 0.45, 5.37 ]
<b>Total (95% CI)</b> Total events: 6 (Upright), 4 Heterogeneity: not applica Test for overall effect: Z =	<b>96</b> 4 (Recumbent) able 5 0.71 (P = 0.48)	100		100.0 %	1.56 [ 0.45, 5.37 ]
			0.1 0.2 0.5 1 2 5 10 Favours upright Favours recumber	nt	

## Analysis 1.21. Comparison I Upright and ambulant positions versus recumbent positions and bed care, Outcome 21 Perinatal mortality.

Review: Maternal positions and mobility during first stage labour

Comparison: I Upright and ambulant positions versus recumbent positions and bed care

Outcome: 21 Perinatal mortality

Study or subgroup	Upright n/N	Recumbent n/N	M-H,Fi	Risk Ratio xed,95% Cl	Risk Ratio M-H,Fixed,95% CI
Bloom 1998	0/536	0/531			0.0 [ 0.0, 0.0 ]
Total (95% CI)	536	531			0.0 [ 0.0, 0.0 ]
Total events: 0 (Upright), 0 (F Heterogeneity: not applicable Test for overall effect: Z = 0.0	Recumbent) 9 0 (P < 0.00001)				
			0.1 0.2 0.5	2 5 10	
			Favours upright	Favours recumbent	
Maternal positions and me	obility during first stage	e labour (Review)			57

# Analysis 2.1. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 1 Duration of first stage of labour: time of epidural insertion to complete cervical dilation (minutes).

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: I Duration of first stage of labour: time of epidural insertion to complete cervical dilation (minutes)

Study or subgroup	Upright		Recumbent			Mea	an Differenc	e	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fixe	ed,95% Cl			IV,Fixed,95% CI
Frenea 2004	25	239 (125)	28	99 (   )		_	-	<b>→</b>	21.1 %	40.00 [ -23.96, 103.96 ]
Collis 1999	110	414 (185)	119	433 (194)			<u> </u>		35.8 %	-19.00 [ -68.09, 30.09 ]
Vallejo 2001	75	240.9 (146.1)	76	211.9 (133.9)		_	-		43.1 %	29.00 [ -15.72, 73.72 ]
Total (95% CI)	210		223			-	-		100.0 %	14.14 [ -15.23, 43.51 ]
Heterogeneity: Chi <sup>2</sup> =	2.80, df =	2 (P = 0.25); I <sup>2</sup> =	-29%							
Test for overall effect:	Z = 0.94 (F	9 = 0.35)								
						1				
					-100	-50	0 50	100		
					Favours	upright	Favours	recumbe	ent	

## Analysis 2.2. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 2 Mode of birth: spontaneous vaginal.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 2 Mode of birth: spontaneous vaginal

Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
Frenea 2004	19/30	23/31		6.2 %	0.85 [ 0.61, 1.20 ]
Vallejo 2001	51/75	56/76	-	15.3 %	0.92 [ 0.75, 1.13 ]
Collis 1999	59/110	64/119	-	16.9 %	1.00 [ 0.78, 1.27 ]
Karraz 2003	117/141	56/74	-	20.2 %	1.10 [ 0.94, 1.27 ]
Nageotte 1997	142/253	150/252	-	41.3 %	0.94 [ 0.81, 1.09 ]
Total (95% CI)	609	552	•	100.0 %	0.97 [ 0.89, 1.06 ]
Total events: 388 (Uprigh	t), 349 (Recumbent)				
Heterogeneity: Chi <sup>2</sup> = 3.4	47, df = 4 (P = 0.48);	l <sup>2</sup> =0.0%			
Test for overall effect: Z =	= 0.58 (P = 0.57)				
	· · · ·				
			0.1 0.2 0.5 1 2 5	5 10	
			Favours recumbent Favours u	ıpright	

Maternal positions and mobility during first stage labour (Review)

## Analysis 2.3. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 3 Mode of birth: operative spontaneous/assisted.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 3 Mode of birth: operative spontaneous/assisted

Study or subgroup	Upright n/N	Recumbent n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
Frenea 2004	6/30	4/31		3.5 %	1.55 [ 0.49, 4.95 ]
Vallejo 2001	10/75	5/76		4.4 %	2.03 [ 0.73, 5.65 ]
Karraz 2003	/ 4	6/74		6.9 %	0.96 [ 0.37, 2.50 ]
Collis 1999	36/110	39/119		33.1 %	1.00 [ 0.69, 1.45 ]
Nageotte 1997	70/253	59/252		52.2 %	1.18 [ 0.88, 1.59 ]
Total (95% CI)	609	552	•	100.0 %	1.16 [ 0.93, 1.44 ]
Total events: 133 (Uprigh	t), 113 (Recumbent)				
Heterogeneity: Chi <sup>2</sup> = 2.	16, df = 4 (P = 0.71);	l <sup>2</sup> =0.0%			
Test for overall effect: Z =	= 1.30 (P = 0.19)				

0.1 0.2 0.5 1 2 5 10 Favours upright Favours recumbent

Maternal positions and mobility during first stage labour (Review)

## Analysis 2.4. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 4 Mode of birth: caesarean section.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 4 Mode of birth: caesarean section

Study or subgroup	Upright	Recumbent	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl		M-H,Fixed,95% Cl
Frenea 2004	5/30	4/31		4.2 %	1.29 [ 0.38, 4.35 ]
Vallejo 2001	14/75	15/76	<b>_</b>	16.0 %	0.95 [ 0.49, 1.82 ]
Collis 1999	15/110	16/119		16.5 %	1.01 [ 0.53, 1.95 ]
Karraz 2003	13/141	12/74		16.9 %	0.57 [ 0.27, 1.18 ]
Nageotte 1997	41/253	43/252		46.3 %	0.95 [ 0.64, 1.40 ]
Total (95% CI)	609	552	•	100.0 %	0.91 [ 0.70, 1.19 ]
Total events: 88 (Upright)	, 90 (Recumbent)				
Heterogeneity: Chi <sup>2</sup> = 2.0	07, df = 4 (P = 0.72)	l <sup>2</sup> =0.0%			
Test for overall effect: Z =	= 0.69 (P = 0.49)				
			0.1 0.2 0.5 1 2 5 10		
			Favours upright Favours recumbent		

## Analysis 2.6. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 6 Maternal pain.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 6 Maternal pain

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
l Requiring additional Bup Nageotte 1997	vivocaine bolus doses 85/253	96/252		100.0 %	0.88 [ 0.70.  .   ]
Total (95% CI)	253	252	•	100.0 %	0.88 [ 0.70, 1.11 ]
Total events: 85 (Treatment), 96 (Control) Heterogeneity: not applicable Test for overall effect: $Z = 1.05$ (P = 0.29)					
			0.01 0.1 10 100 Favours treatment Favours contro	I	

Maternal positions and mobility during first stage labour (Review)

## Analysis 2.7. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 7 Analgesia type.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 7 Analgesia type

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI	M-H,Fixed,95% CI
l Opioid				
2 Epidural				
Fernando 1994	20/20	20/20		0.0 [ 0.0, 0.0 ]
Frenea 2004	30/30	31/31		0.0 [ 0.0, 0.0 ]
Karraz 2003	4 / 4	74/74		0.0 [ 0.0, 0.0 ]
Nageotte 1997	253/253	252/252		0.0 [ 0.0, 0.0 ]
Vallejo 2001	75/75	76/76		0.0 [ 0.0, 0.0 ]

0.1 0.2 0.5 1 2 5 10

Favours upright Favours recumbent

## Analysis 2.8. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 8 Analgesia amount.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 8 Analgesia amount

Study or subgroup	Upright	Recumbent			Mean Difference	Mean Difference Weight		
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fixed,95% CI		IV,Fixed,95% CI	
l Bupivocaine								
Karraz 2003	117	27 (11)	56	23 (  )	+	3.3 %	4.00 [ 0.50, 7.50 ]	
Frenea 2004	30	6.4 (2.2)	31	8.4 (3.6)	•	18.2 %	-2.00 [ -3.49, -0.51 ]	
Collis 1999	110	6.74 (2.74)	9	7.43 (3.46)	•	62.6 %	-0.69 [ -1.50, 0.12 ]	
Subtotal (95% CI)	257		206			84.2 %	-0.79 [ -1.48, -0.09 ]	
Heterogeneity: $Chi^2 = 9.7$	7, df = 2 (P	= 0.0 l ); l <sup>2</sup> =80%						
Test for overall effect: Z =	2.23 (P = 0	.026)						
2 Ropivacaine								
Vallejo 2001	75	8.7 (70.1)	76	99 (45.9)		0.1 %	19.70 [ 0.77, 38.63 ]	
Subtotal (95% CI)	75		76		•	0.1 %	19.70 [ 0.77, 38.63 ]	
Heterogeneity: not applica	able							
Test for overall effect: Z =	2.04 (P = 0	.04I)						
3 Fentanyl								
Collis 1999	110	17.38 (6.41)	119	17.76 (5.97)	•	15.7 %	-0.38 [ -1.99, 1.23 ]	
Subtotal (95% CI)	110		119			15.7 %	-0.38 [ -1.99, 1.23 ]	
Heterogeneity: not applica	able							
Test for overall effect: Z =	0.46 (P = 0	.64)						
Total (95% CI)	442		401			100.0 %	-0.70 [ -1.34, -0.06 ]	
Heterogeneity: Chi <sup>2</sup> = 14	.45, df = 4 (F	$P = 0.01$ ); $ ^2 = 72\%$						
Test for overall effect: Z =	2.16 (P = 0	.031)						
Test for subgroup differen	ces: Chi <sup>2</sup> = 4	4.68, df = 2 (P = 0.	10), I <sup>2</sup> =57%					

-100 -50

Favours upright

0

50 100

Favours recumbent

Maternal positions and mobility during first stage labour (Review)

## Analysis 2.9. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 9 Duration of second stage of labour (minutes).

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 9 Duration of second stage of labour (minutes)

Study or subgroup	Upright	I	Recumbent			Mear	Differenc	ce	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)		IV,Fixed	1,95% CI			IV,Fixed,95% CI
Frenea 2004	25	56 (42)	28	62 (59)					41.2 %	-6.00 [ -33.36, 21.36 ]
Vallejo 2001	75	97.3 (76)	76	89.1 (67.3)		-	-		58.8 %	8.20 [ -14.71, 31.11 ]
Total (95% CI)	100		104			-	•		100.0 %	2.35 [ -15.22, 19.91 ]
Heterogeneity: Chi <sup>2</sup> =	= 0.61, df = 1	$(P = 0.44); I^2 = 0.05$	%							
Test for overall effect:	Z = 0.26 (P =	= 0.79)								
					1		1			
					-100	-50 0	50	100		
					Favours	upright	Favours	recumb	ent	

## Analysis 2.10. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 10 Augmentation of labour using oxytocin.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 10 Augmentation of labour using oxytocin

Study or subgroup	Upright	Recumbent		R	Risk Ratio		Weight	Risk Ratio
	n/N	n/N		M-H,Fix	ed,95% Cl			M-H,Fixed,95% Cl
Frenea 2004	24/30	27/31			_		7.5 %	0.92 [ 0.73, 1.15 ]
Vallejo 2001	27/75	31/76					8.7 %	0.88 [ 0.59, 1.32 ]
Karraz 2003	65/141	35/74			-		13.0 %	0.97 [ 0.72, 1.32 ]
Collis 1999	75/110	83/119		-	F		22.5 %	0.98 [ 0.82, 1.16 ]
Nageotte 1997	173/253	171/252		-	-		48.4 %	1.01 [ 0.89, 1.14 ]
Total (95% CI)	609	552		•			100.0 %	0.98 [ 0.90, 1.07 ]
Total events: 364 (Uprigh	t), 347 (Recumbent)							
Heterogeneity: $Chi^2 = 0.7$	79, df = 4 (P = 0.94);	l <sup>2</sup> =0.0%						
Test for overall effect: Z =	= 0.47 (P = 0.64)							
			0.2	0.5	2	5		
			Favour	s upright	Favours 1	recumbent		

## Analysis 2.12. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 12 Hypotension requiring intervention.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 12 Hypotension requiring intervention

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/IN	n/IN	MI-H,FIXEd,95% CI	I*I-H,FIXed,95% CI
Karraz 2003	0/141	0/74		0.0 [ 0.0, 0.0 ]
Nageotte 1997	2/253	4/252	• <b>——</b>	0.50 [ 0.09, 2.69 ]
Frenea 2004	9/30	6/31		1.55 [ 0.63, 3.82 ]
Total (95% CI)	424	357		1.12 [ 0.52, 2.45 ]
Total events: 11 (Upright), 10	0 (Recumbent)			
Heterogeneity: Chi <sup>2</sup> = 1.38,	df =   (P = 0.24); $ ^2 = 27\%$			
Test for overall effect: $Z = 0$ .	.30 (P = 0.77)			
			0.1 0.2 0.5 1 2 5 10	
			Favours upright Favours recumbent	

## Analysis 2.17. Comparison 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women), Outcome 17 Apgar scores.

Review: Maternal positions and mobility during first stage labour

Comparison: 2 Upright and ambulant positions versus recumbent positions or bedcare (with epidural: all women)

Outcome: 17 Apgar scores

Study or subgroup	Upright	Recumbent	Risk Ratio	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	M-H,Fixed,95% Cl
Apgar < 7 at   min				
Vallejo 2001	5/75	5/76		1.01 [ 0.31, 3.36 ]
Subtotal (95% CI)	75	76	-	1.01 [ 0.31, 3.36 ]
Total events: 5 (Upright), 5 (Recu	umbent)			
Heterogeneity: not applicable				
Test for overall effect: $Z = 0.02$ (	P = 0.98)			
2 Apgar < 7 at 5 mins				
Fernando 1994	0/20	0/20		0.0 [ 0.0, 0.0 ]
Frenea 2004	0/30	0/31		0.0 [ 0.0, 0.0 ]
Collis 1999	2/110	1/119		2.16 [ 0.20, 23.53 ]
Nageotte 1997	1/253	2/252		0.50 [ 0.05, 5.46 ]
Subtotal (95% CI)	413	422	-	1.04 [ 0.21, 5.05 ]
Total events: 3 (Upright), 3 (Recu	umbent)			
Heterogeneity: $Chi^2 = 0.73$ , df =	I (P = 0.39); I <sup>2</sup> =0.0%			
Test for overall effect: $Z = 0.05$ (	P = 0.96)			
Total (95% CI)	488	498	+	1.02 [ 0.39, 2.66 ]
Total events: 8 (Upright), 8 (Recu	umbent)			
Heterogeneity: $Chi^2 = 0.73$ , df =	2 (P = 0.70); l <sup>2</sup> =0.0%			
Test for overall effect: $Z = 0.05$ (	P = 0.96)			
			0.01 0.1 1 10 100	
			Favours upright Favours recumb	ent

## Analysis 3.1. Comparison 3 Standing and walking versus non-ambulant positions, Outcome I Duration of the first stage of labour.

Review: Maternal positions and mobility during first stage labour

Comparison: 3 Standing and walking versus non-ambulant positions

Outcome: I Duration of the first stage of labour

Study or subgroup	Ambulant Non ambulant				Mean Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Random,95% CI		IV,Random,95% CI
I Studies where ambulati	on was encour	raged					
Haukkama 1982	13	10.2 (5.4)	12	8.9 (4.6)		4.7 %	1.30 [ -2.62, 5.22 ]
McManus 1978	10	10.5 (3.7)	10	10.5 (4.4)	<b>+</b>	5.4 %	0.0 [ -3.56, 3.56 ]
Williams 1980	25	7.9 (4.9)	30	7.4 (3.2)		9.5 %	0.50 [ -1.74, 2.74 ]
Andrews 1990	20	3.9 (1.5)	20	5.41 (1.5)	-	15.8 %	-1.51 [ -2.44, -0.58 ]
Bloom 1998	272	7.6 (3.9)	272	7.3 (3.9)	+	17.0 %	0.30 [ -0.36, 0.96 ]
Subtotal (95% CI) Heterogeneity: $Tau^2 = 0.4$ Test for overall effect: Z =	<b>340</b> 86; Chi <sup>2</sup> = 10.8 = 0.34 (P = 0.7	80, df = 4 (P = 0.0 73)	<b>344</b> 3); I <sup>2</sup> =63%		•	52.5 %	-0.20 [ -1.36, 0.96 ]
2 Studies where women	were not amb	ulant					
Chen 1987	22	3.25 (2.25)	38	4.23 (2.5)		14.3 %	-0.98 [ -2.21, 0.25 ]
Phumdoung 2007	40	3.54 (1.91)	43	6.33 (2.1)	+	16.1 %	-2.79 [ -3.65, -1.93 ]
Mitre 1974	50	5.47 (1.71)	50	7.25 (1.64)	-	17.0 %	-1.78 [ -2.44, -1.12 ]
Subtotal (95% CI)	112		131		•	47.5 %	-1.92 [ -2.83, -1.01 ]
Heterogeneity: $Iau^2 = 0.4$	$43; Chi^2 = 6.28$	8, df = 2 (P = $0.04$	); l² =68%				
lest for overall effect: $\angle =$	= 4.13 (P = 0.0 450	00037)	475		•	100 0 0/	
<b>Iotal (95% CI)</b>	<b>452</b>		4/3		•	100.0 %	-0.9/ [ -1.96, 0.02 ]
Test for some loft of a former of the second	38; Cni~ - 40.8	85, df — 7 (P<0.00)	JUT); I~ -83%				
lest for overall effect: ∠ =	= 1.93 (P = 0.0	J54)					
				1		1	
				- I C	) -5 0 5 I	0	

Favours upright

Favours recumbent

HISTORY

Protocol first published: Issue 4, 2002

Review first published: Issue 2, 2009

11 November 2008 Amended Converted to new review format.

## CONTRIBUTIONS OF AUTHORS

The text of the review was drafted by Lucy Lewis, Therese Dowswell and Annemarie Lawrence. Justus Hofmeyr and Cathy Styles commented on drafts. Data extraction and data entry were carried out by Lucy Lewis, Therese Dowswell and Annemarie Lawrence.

## DECLARATIONS OF INTEREST

None known.

## SOURCES OF SUPPORT

#### Internal sources

- Griffith University, School of Nursing, Nathan Campus, Queensland, Australia.
- Centre for Clinical Studies Women's and Children's Health, Mater Hospital, South Brisbane, Queensland, Australia.
- The University of Liverpool, UK.
- University of Adelaide, Australian Research Centre for Health of Women and Babies, Women's and Children's Hospital, Adelaide, South Australia, Australia.
- James Cook University, School of Midwifery and Nutrition, Townsville, Queensland, Australia.
- University of Queensland, School of Nursing and Midwifery, Royal Brisbane and Women's Hospital, Brisbane, Queensland, Australia.

#### **External sources**

- Department of Health and Ageing, Commonwealth Government, Canberra ACT, Australia.
- National institute for Health Research, UK. NIHR NHS Cochrane Collaboration Programme Grant Scheme award for NHS-prioritised centrally-managed, pregnancy

and childbirth systematic reviews: CPGS02

## DIFFERENCES BETWEEN PROTOCOL AND REVIEW

The methods section has been updated to reflect changes in methods and software.

## INDEX TERMS

#### Medical Subject Headings (MeSH)

Labor Stage, First [\*physiology]; Posture [\*physiology]; Randomized Controlled Trials as Topic; Supine Position [physiology]; Time Factors; Walking [\*physiology]

## MeSH check words

Female; Humans; Pregnancy